\documentclass[11pt,dvips]{article}
\textheight = 24truecm \textwidth = 17truecm \hoffset = -2.3truecm
\voffset = -2truecm
\renewcommand{\baselinestretch}{1.2}
\usepackage{graphicx}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{latexsym}
\usepackage{color}
\begin{document}
\begin{thebibliography}{99}


\bibitem{NC-QIQC-2000} M.A. Nielsen and I.L. Chuang, {\it Quantum Computation and
Quantum Information} (Cambridge Univ. Press, Cambridge, 2000).

\bibitem{Vedral-RMP-2002} V. Vedral, Rev. Mod. Phys. {\bf 74} (2002)
197.


\bibitem{Horodecki-RMP-2009} R. Horodecki, P. Horodecki, M. Horodecki and K.
Horodecki, Rev. Mod. Phys. {\bf 81}(2009) 865.

\bibitem{Guhne} O. G\"uhne and G. T\'oth, Phys. Rep. {\bf 474} (2009) 1.

\bibitem{vedral} K. Modi, A. Brodutch, H. Cable, T. Paterek and V. Vedral,
 Rev. Mod. Phys. {\bf 84} (2012) 1655.

%\bibitem{vedral} K. Modi, A. Brodutch, H. Cable, T. Paterek and V. Vedral, {\rm Quantum discord and other measures of quantum
%correlation}, {\tt arXiv:1112.6238}.

\bibitem{Rungta}  P. Rungta, V. Buzek, C.M. Caves, M. Hillery and G.J. Milburn, Phys. Rev. A {\bf 64} (2001) 042315.

\bibitem{Ben3}  C.H. Bennett, D.P. DiVincenzo, J. Smolin and W.K.
Wootters, Phys. Rev. {\bf A 54} (1996) 3824.

\bibitem{Wootters}  W.K. Wootters, Phys. Rev. Lett. {\bf 80} (1998) 2245.

\bibitem{Coffman}  V. Coffman, J. Kundu and W.K. Wootters, Phys. Rev.
A {\bf 61} (2000) 052306.

\bibitem{Vedral-et-al} L. Henderson and V. Vedral, J. Phys. A {\bf 34}(2001)  6899;
V. Vedral, Phys. Rev. Lett. {\bf 90}  (2003) 050401; J. Maziero, L.
C. Cel\'eri, R.M. Serra and V. Vedral, Phys. Rev A {\bf 80} (2009)
044102.

\bibitem{Ollivier-PRL88-2001} H. Ollivier and W.H. Zurek, Phys. Rev. Lett. {\bf 88} (2001) 017901.

\bibitem{Dakic2010} B. Dakic, V. Vedral and C. Brukner, phys. Rev. Lett. {\bf 105} (2010) 190502.

\bibitem{Bellomo1}
B. Bellomo,  R. Lo Franco and G. Compagno, Phys. Rev. A 86, 012312
(2012).

\bibitem{Bellomo2}
B. Bellomo, G.L. Giorgi, F. Galve, R. Lo Franco, G. Compagno and R.
Zambrini, Phys. Rev. A {\bf 85} (2012) 032104.

\bibitem{Dajka} J. Dajka, J. Luczka and P. H\"anggi,
Phys. Rev. A {\bf 84} (2011) 032120.

\bibitem{Paula1} J.D. Montealegre, F.M. Paula, A. Saguia and M.S. Sarandy,
Phys. Rev. A {\bf 87} (2013) 042115.

\bibitem{Aaronson} B. Aaronson, R. Lo Franco, G. Compagno and G. Adesso,
New Journal of Physics {\bf 15} (2013) 093022.

\bibitem{Paula2} F.M. Paula, T.R. de Oliveira and M.S. Sarandy,
Phys. Rev. A {\bf 87} (2013) 064101.

\bibitem{wigner} E. P. Wigner and M. M. Yanasse, Proc. Natl. Acad. Sci. USA \textbf{49}, 910 (1963).

\bibitem{girolami} D. Girolami, T. Tufarelli, and G. Adesso, Phys. Rev. Lett. \textbf{110}, 240402 (2013).

\bibitem{luo} S. Luo, Phys. Rev. Lett. \textbf{91}, 180403 (2003).

\bibitem{petz} D. Petz and C. Ghinea, arXiv:1008.2417 (2010).

\bibitem{luo2} S. Luo, Proc. Amer. Math. Soc. \textbf{132}, 885 (2003).

\bibitem{Karimipour} A. Mani, V. Karimipour and L. Memarzadeh,
Phys. Rev. A   {\ bf  91}  (2015) 012304

\bibitem{Sen} A. Sen, A. Bhar  and D. Sarkar, Quant. Inf. Proc. {\bf 14} (2015) 269.

\bibitem{solano} T. Bastin, S. Krins, P. Mathonet, M.  Godefroid, L.  Lamata and E. Solano, Phys. Rev. Lett. {\bf 103} (2009) 070503.

\bibitem{mixed} T. Bastin, P.  Mathonet and E. Solano, Operational Entanglement Families of Symmetric Mixed N-qubit states, quant-ph/arxiv:1011.1243

\bibitem{usa1}  A.R. Usha Devi, Sudha and A.K. Rajagopal,   Determining the whole pure symmetric N-qubit state from its parts,  quant-ph/arXiv:1003.2450.

\bibitem{usa2} A.R. Usha Devi, Sudha and A.K. Rajagopal, Interconvertibility and irreducibility of permutation symmetric three qubit pure states,
quant-ph/arXiv:1002.2820.

\bibitem{markham1} D. Markham, Entanglement and symmetry in permutation symmetric states, quant-ph/arxiv: 1001.0343.

\bibitem{markham2} M. Aulbach, D. Markham and M.  Murao, New. J. Phys., {\bf 12}  (2010) 073025.

\bibitem{gebastin} J. Martin, O. Giraud, P.A. Braun, D. Braun and T.  Bastin, Phys. Rev. A {\bf 81}  (2010) 062347.

\bibitem{markham3} M. Aulbach, D. Markham and M.  Murao, Geometric Entanglement of Symmetric States and the Majorana Representation,  quant-ph/arxiv: 1010.4777

\bibitem{Maj32}  E. Majorana, Nuovo Cimento {\bf 9} (1932) 43.

\bibitem{Dic54}  R.H. Dicke, Phys. Rev. {\bf 93} (1954) 99.

\bibitem{wang1} X. Wang and  K. M{\o}lmer, Eur. Phys. J.  D {\bf 18} (2002) 385.

\bibitem{Retzker}  A. Retzker, E. Solano, and B. Reznik, Phys. Rev. A {\bf 75} (2007) 022312.
; I. E. Linington and N. V. Vitanov, Phys. Rev. A {\bf 77} (2008) 062327
;C. E. L\'{o}pez, J. C. Retamal and E. Solano, Phys. Rev. A {\bf 76} (2007) 033413.


\bibitem{Stockton}  J. K. Stockton, J. M.
Geremia, A. C. Doherty and H. Mabuchi, Phys. Rev. A {\bf 67} (2003) 022112
; W. D\"ur, Phys. Rev. A {\bf 63} (2001) 020303 R.


\bibitem{daoud2} M. Daoud, R. Ahl Laamara and W. Kaydi,
J. Phys. A: Math. Theor. {\bf 46} (2013) 395302.


\bibitem{GHZ} D. M. Greenberger, M.A.  Horne and  A. Zeilinger, Phys. Rev. Lett {\bf 75} (1995) 2064.

\bibitem{Dur00} W. D\"ur, G. Vidal and J.I. Cirac, Phys. Rev. A
{\bf 62} (2000) 062314


\bibitem{Dicke} R.H. Dicke, Phys. Rev. {\bf 93} (1954) 99.


\bibitem{LMG} H.J. Lipkin, N. Meshkov and A.J. Glick, Nucl. Phys. {\bf 62} (1965) 188.


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%














\end{thebibliography}

\thispagestyle{empty}
\begin{center}

\vspace{1.8cm}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
 {\large {\bf Pairwise local quantum uncertainty in superpositions of Dicke states }}\\
 %%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\vspace{1.5cm} {\bf xxxxxx}$^{a,b,c}${\footnote { email: {\sf
m$_{-}$daoud@hotmail.com}}}, {\bf yyyyyy}$^{d,e}$ {\footnote {
email: {\sf ahllaamara@gmail.com}}}  and {\bf zzzzzz}$^{d}$
{\footnote { email: {\sf ????????}}} \\
\vspace{0.5cm}
$^{a}${\it Department of Physics, Faculty of Sciences-Ain Chock, University Hassan II Casablanca, Morocco}\\[0.5em]
$^{b}${\it Abdus Salam  International Centre for Theoretical Physics, Trieste, Italy}\\[0.5em]
$^{c}${\it Department of Physics, Faculty of Sciences,  University Ibnou Zohr, Agadir , Morocco}\\[0.5em]
$^{d}${\it LPHE-Modeling and Simulation, Faculty  of Sciences, Rabat, Morocco}\\[0.5em]
$^{e}${\it Centre of Physics and Mathematics,  Rabat, Morocco}\\[0.5em]



\vspace{3cm} {\bf Abstract}
\end{center}
\baselineskip=18pt
\medskip


We give a closed expression of local quantum uncertainty for
two-qubit $X$ states. A detailed analysis is performed for multi-partite spin systems for which we give
the pairwise local quantum uncertainty.





\vspace{1cm}








\newpage
\section{Introduction}




Quantum correlations in multipartite systems are a fundamental
resource in various protocols of quantum information processing  \cite{NC-QIQC-2000,Vedral-RMP-2002,Horodecki-RMP-2009,Guhne}. In
this respect, the characterization the degree of  quantumness  of
correlations between the different parts of a composite system is
highly desirable. During the last two decades, several quantifiers
were investigated in the literature (for a recent review see
\cite{vedral}). The most utilized
ones to capture the main features of quantum correlations are the
concurrence, entanglement of formation, quantum discord and its
different geometric versions \cite{Rungta,Ben3,Wootters,Coffman,Vedral-et-al,Ollivier-PRL88-2001}. The interest in quantum discord lies
in the existence of nonclassical correlations even in separable
states \cite{Vedral-et-al,Ollivier-PRL88-2001}. The entanglement of formation  does
not account for all nonclassical aspects of correlations and
unentangled mixed states can possess quantum correlations. This
explains the important number of works devoted to the significance
and the computation of quantum discord in different quantum systems.
However, the derivation of the explicit expression of quantum
discord of an arbitrary quantum system is very challenging in
general. An alternative way to overcome this problem consists in
utilizing geometric methods to quantify the distance between a
bipartite state and its closed one encompassing only classical
correlations \cite{Dakic2010,Bellomo1,Bellomo2}
(see also \cite{Dajka,Paula1,Aaronson,Paula2}).


Recently, it was shown that the concept of quantum discord is deeply
related to the formalism of quantum uncertainty on local
observables. This kind of quantum correlations quantifier  uses the
skew information, introduced in Ref. \cite{wigner}, to determine the
uncertainty in the measurement of an observable. More precisely, the
local quantum uncertainty is given by the minimum of the skew
information over all possible observables. This measure leads to the
closed analytical expression characterizing the quantum correlations
for any qubit-qudit bipartite system \cite{girolami}.  It must be
noticed that this quantum correlation measure is related to quantum
Fisher information \cite{luo,petz,luo2} which is the key ingredient
in estimating precisions  in quantum metrology
protocols~\cite{girolami}. It also quantifies the speed of
evolution of a quantum system \cite{girolami}.


Taking advantages from the recent progress in understanding this
kind of quantum correlations quantifier, we give in this paper the
explicit analytical expressions of local quantum uncertainty for a
two qubit $X$ states including some special states for which the
local quantum uncertainty was recently derived
\cite{Karimipour,Sen}.  As illustration we consider
the pairwise quantum correlation in multipartite symmetric qubit
states.






The paper is structured as follows. In the following section, we introduce
the concept of local quantum uncertainty. The explicit anlaytical for of this  measure of discord-like
quantum correlations is derived for an arbitrary two-qubit $X$ states.  To illustrate our purpose, we consider collective
$n$ qubit systems  possessing
parity and exchange symmetries.  A special attention is devoted to the evaluation of pairwise
local quantum uncertainty in Dicke states, even and odd spin coherent states and finally Kitagawa-Ueda system.
 Concluding remarks close this
paper.

\section{Local quantum uncertainty in two qubit $X$ states}


\subsection{Local quantum uncertainty: Definition}

The concept of local quantum uncertainty is considered as a faithful
quantifier of quantum correlation. It quantifies the minimal quantum
uncertainty in a quantum state due to a measurement of a local
observable \cite{girolami}. For a quantum state $\rho_{AB}$, the
local quantum uncertainty is defined as
\begin{equation}
\mathcal{U}(\rho_{AB}) \equiv \min_{K} \mathcal{I}(\rho_{AB}, K \otimes
I), \label{LQU}
\end{equation}
where $K$ is some local observable on the subsystem $A$ and the quantity
\begin{equation}
\mathcal{I}(\rho_{AB},  K)=-\frac{1}{2}{\rm
Tr}([\sqrt{\rho_{AB}},K]^{2})
\end{equation}
 is the skew information \cite{wigner,luo}. The skew information
represents the non-commutativity between the state and the
observable $K$.
The analytical evaluation the local quantum
uncertainty requires a
 minimization procedure  over the set of all
observales acting on the part $A$. A closed form for qubit-qudit
systems was derived in \cite{girolami}. In particular, for
qubits (spin-$\frac{1}{2}$ particles),  it is given by  \cite{girolami}
\begin{equation}
 \mathcal{U}(\rho) = 1 - \lambda_{max}\{W\},
\end{equation}
where  $\lambda_{max}$ denotes the maximal  eigenvalue of the matrix
$W$ whose matrix elements are given by
\begin{equation}\label{matrixomega}
 (W)_{ij} \equiv  {\rm
Tr}\{\sqrt{\rho_{AB}}(\sigma_{i}\otimes
{\bf I})\sqrt{\rho_{AB}}(\sigma_{j}\otimes {\bf I})\},
\end{equation}
The local quantum uncertainty is zero when the observable and the
state of the system commute. This  indicates the absence of any
quantum correlation between the components of the bipartite system.
The local quantum uncertainty can be considered  as quantum discord
measure. In this sense, it satisfies all necessary requirements.
Indeed, local quantum uncertainty vanishes for all zero quantum discord states. It
is also invariant under local unitary operations on un-measured
qubit. In the following we derive the explicit form of local quantum
uncertainty in  two-qubit $X$ states.

\subsection{Local quantum uncertainty for $X$ states}

We consider a two-qubit system which is described by an $X$-shaped
mixed state.  Two-qubit $X$ states include various types of quantum
states usually used in investigating entanglement and quantum
correlations in various condensed matter models such ones describing
spin collective systems. In the computational basis of a two qubit
system,  the $X$ density matrices have non-zero entries only along
the diagonal and anti-diagonal and therefore they are parameterized
by seven real parameters. The corresponding symmetry is fully
characterized by the $su(2) \times su(2) \times u(1)$ subalgebra of
the full $su(4)$ algebra describing an arbitrary two-qubit system.
Many analytical calculations of concurrence, entanglement of
formation, quantum discord  can be carried  out easily for $X$
states leading to interesting results in studying their properties
and especially their evolution under dissipative processes
\cite{ref6,ref7}. The density matrix for a two-qubit $X$ state writes as
\begin{equation}
\rho =  \left(
\begin{array}{cccc}
\rho_{11} & 0 & 0 & \rho_{14} \\
0 & \rho_{22} & \rho_{23} & 0 \\
0 & \rho_{32} & \rho_{33} & 0 \\
\rho_{41} & 0 & 0 & \rho_{44}
\end{array}
\right). \label{eqn1}
\end{equation}
in the computational basis $\{ \vert 00 \rangle, \vert 01\rangle,
\vert 10\rangle, \vert 11\rangle\}$. The entries are subjected to
the normalization property (${\rm Tr} \rho= 1$), the
positivity condition ($\rho_{11}\rho_{44} \geq \vert \rho_{14}
\vert^2$ and $\rho_{22}\rho_{33} \geq \vert \rho_{23} \vert^2$) and
 the complex conjugation requirement  ($\rho_{14} = \rho^{\star}_{14}$ and $\rho_{23} = \rho^{\star}_{32}$ ). The phase factors
$e^{i\theta_{14}}= \frac{\rho_{14}}{\vert \rho_{14}\vert }$ and
$e^{i\theta_{23}}= \frac{\rho_{23}}{\vert \rho_{23}\vert }$
 of the off diagonal elements can be removed using the local unitary
transformations
$$ \vert 0 \rangle_1 \rightarrow \exp \bigg(-\frac{i}{2} ( \theta_{14}+ \theta_{23})\bigg) \vert 0 \rangle_1 \quad  \vert 0 \rangle_2 \rightarrow \exp \bigg(-\frac{i}{2} (\theta_{14}-\theta_{23} )\bigg) \vert 0 \rangle_2$$
 and  the
rank of the density matrix $\rho$ remains unchanged. Thus, all
entries of the density matrix can be taken positive.  The
eigenvalues of the density matrix $\rho$ write
$$\lambda_{1} = \frac{1}{2} t_1 + \frac{1}{2} \sqrt{t_1^2 - 4 d_1}, \quad\lambda_{2} = \frac{1}{2} t_2 + \frac{1}{2} \sqrt{t_2^2 - 4 d_2},
\quad \lambda_{3} = \frac{1}{2} t_2 - \frac{1}{2} \sqrt{t_2^2 - 4
d_2}, \quad\lambda_{4} = \frac{1}{2} t_1 - \frac{1}{2} \sqrt{t_1^2 -
4 d_1} $$ with $ t_1 = \rho_{11}  + \rho_{44}, \quad t_2= \rho_{22}
+ \rho_{33},\quad d_1 = \rho_{11}\rho_{44} - \rho_{14}\rho_{41},
{\rm and}\quad d_2= \rho_{22}\rho_{33}- \rho_{32}\rho_{23} $. The
Fano-Bloch decomposition of the state $\rho$ writes as
\begin{equation}\label{fano-bloch-rho}
\rho = \frac{1}{4}\sum_{\alpha, \beta} {R}_{\alpha \beta} \sigma_{\alpha} \otimes \sigma_{\beta}
\end{equation}
where the correlation matrix ${R}_{\alpha \beta}$ are given by $ {
R}_{\alpha \beta} = {\rm Tr} (\sqrt{\rho}~\sigma_{\alpha} \otimes
\sigma_{\beta} )$. They write
%$$ R_{03} = 1 -2  \rho_{22} - 2\rho_{44},\quad  R_{30} =    1-  2\rho_{33} - 2\rho_{44},\quad  R_{11} = 2~{\rm Re}(\rho_{32} + \rho_{41}),\quad  R_{22} =  2~{\rm Re}(\rho_{32} - \rho_{41})$$
%$$ R_{12} = -2i~{\rm Im}(\rho_{41} - \rho_{32})\quad  R_{21} =  -2i~{\rm Im}(\rho_{41} + \rho_{32}),\quad R_{00} = \rho_{11}+\rho_{22}+\rho_{33}+\rho_{44}= 1 ,\quad  R_{33} =    1-  2\rho_{22} - 2\rho_{33}.$$
$$ R_{03} = 1 -2  \rho_{22} - 2\rho_{44},\quad  R_{30} =    1-  2\rho_{33} - 2\rho_{44},\quad  R_{11} = 2~{\rm Re}(\rho_{32} + \rho_{41}),$$
$$  R_{22} =  2~{\rm Re}(\rho_{32} - \rho_{41}) ,\quad R_{00} = \rho_{11}+\rho_{22}+\rho_{33}+\rho_{44}= 1 ,\quad  R_{33} =    1-  2\rho_{22} - 2\rho_{33}.$$
For nonzero values of $t_1$ and $t_2$, it is simple to check that
the square root of the density matrix $\rho$ writes
\begin{equation}
\sqrt{\rho} =  \left(
\begin{array}{cccc}
\frac{\rho_{11}+\sqrt{d_1}}{\sqrt{t_1 + 2\sqrt{d_1}}} & 0 & 0 & \frac{\rho_{14}}{\sqrt{t_1 + 2\sqrt{d_1}}}\\
0 & \frac{\rho_{22}+\sqrt{d_2}}{\sqrt{t_2 + 2\sqrt{d_2}}} & \frac{\rho_{23}}{\sqrt{t_2 + 2\sqrt{d_2}}} & 0 \\
0 & \frac{\rho_{32}}{\sqrt{t_2 + 2\sqrt{d_2}}} & \frac{\rho_{33}+\sqrt{d_2}}{\sqrt{t_2 + 2\sqrt{d_2}}} & 0 \\
\frac{\rho_{41}}{\sqrt{t_1 + 2\sqrt{d_1}}} & 0 & 0 &
\frac{\rho_{44}+\sqrt{d_1}}{\sqrt{t_1 + 2\sqrt{d_1}}}
\end{array}
\right). \label{sqrtrho}
\end{equation}
The eigenvalues $\sqrt{\lambda_{1}}$, $\sqrt{\lambda_{2}}$,
$\sqrt{\lambda_{3}}$ and $\sqrt{\lambda_{4}}$ of the matrix
$\sqrt{\rho} $ can be rewritten as
$$\sqrt{\lambda_{1}} = \frac{1}{2}  \sqrt{t_1 + 2 \sqrt{d_1}}   + \frac{1}{2} \sqrt{t_1 - 2 \sqrt{d_1}}, \qquad \sqrt{\lambda_{2}} = \frac{1}{2} \sqrt{t_2 + 2 \sqrt{d_2}}  + \frac{1}{2} \sqrt{t_2 - 2 \sqrt{d_2}} $$
$$\sqrt{\lambda_{3}}  = \frac{1}{2} \sqrt{t_2 + 2 \sqrt{d_2}} - \frac{1}{2} \sqrt{t_2 - 2 \sqrt{d_2}}, \qquad \sqrt{\lambda_{4}} = \frac{1}{2} \sqrt{t_1 + 2 \sqrt{d_1}} - \frac{1}{2} \sqrt{t_1 - 2 \sqrt{d_1}} $$
In Fano-Bloch representation, the matrix $\sqrt{\rho}$ writes as
$$ \sqrt{\rho} = \frac{1}{4}\sum_{\alpha, \beta} {\cal R}_{\alpha \beta} \sigma_{\alpha} \otimes \sigma_{\beta}$$
with $ {\cal R}_{\alpha \beta} = {\rm Tr}
(\sqrt{\rho}~\sigma_{\alpha} \otimes \sigma_{\beta} )$. The non
vanishing matrix correlation elements ${\cal R}_{\alpha \beta}$ are
explicitly given by
$$  {\cal R}_{00} =  \sqrt{t_1 + 2 \sqrt{d_1}}  +  \sqrt{t_2 + 2 \sqrt{d_2}}   \qquad  {\cal R}_{03} =   \frac{1}{2} \frac{R_{30}  + R_{03}  }{\sqrt{t_1 + 2\sqrt{d_1}}} -  \frac{1}{2} \frac{R_{30}- R_{03} }{\sqrt{t_2 + 2\sqrt{d_2}}}$$
$$  {\cal R}_{30} = \frac{1}{2} \frac{R_{30}  + R_{03}  }{\sqrt{t_1 + 2\sqrt{d_1}}} + \frac{1}{2} \frac{R_{30}- R_{03} }{\sqrt{t_2 + 2\sqrt{d_2}}} \qquad {\cal R}_{11} = \frac{1}{2} \frac{R_{11}  + R_{22}  }{\sqrt{t_2 + 2\sqrt{d_2}}} +  \frac{1}{2} \frac{R_{11}- R_{22} }{\sqrt{t_1 + 2\sqrt{d_1}}}$$
%$$  {\cal R}_{12} =  \frac{1}{2} \frac{R_{12}  + R_{21}  }{\sqrt{t_1 + 2\sqrt{d_1}}} +  \frac{1}{2} \frac{R_{12}- R_{21} }{\sqrt{t_2 + 2\sqrt{d_2}}} \qquad  {\cal R}_{21} = \frac{1}{2} \frac{R_{12}  + R_{21}  }{\sqrt{t_1 + 2\sqrt{d_1}}} -  \frac{1}{2} \frac{R_{12}- R_{21} }{\sqrt{t_2 + 2\sqrt{d_2}}}$$
$$  {\cal R}_{22} = \frac{1}{2} \frac{R_{11}  + R_{22}  }{\sqrt{t_2 + 2\sqrt{d_2}}} -  \frac{1}{2} \frac{R_{11}- R_{22} }{\sqrt{t_1 + 2\sqrt{d_1}}} \qquad  {\cal R}_{33} =  \sqrt{t_1 + 2 \sqrt{d_1}}  -  \sqrt{t_2 + 2 \sqrt{d_2}} $$
At this stage, we have the tools to evaluate the matrix elements
defined by
$$ \omega_{ij} = {\rm Tr} \bigg(\sqrt{\rho}~(\sigma_{i} \otimes \sigma_{0}) \sqrt{\rho}~(\sigma_{j} \otimes \sigma_{0}) \bigg)$$
where $i$ and $j$ take the values 1, 2, 3. Using the following
identities
$$\{ \sigma_i , \sigma_j\} = 2\delta_{ij}\quad  {\rm Tr}(\sigma_i\sigma_j)= 2\delta_{ij} \quad {\rm Tr}(\sigma_i\sigma_j\sigma_k\sigma_l)=
2(\delta_{ij}\delta_{kl}- \delta_{ik}\delta_{jl}+
\delta_{il}\delta_{jk}),$$ one shows that the matrix $W$ is diagonal
and the diagonal elements are
\begin{eqnarray} \label{omegaii}
\omega_{ii} = \frac{1}{4} \bigg[ \sum_{\beta}\bigg( {\cal
R}^2_{0\beta} - \sum_k {\cal R}^2_{k\beta} \bigg) \bigg] +
\frac{1}{2}\sum_{\beta} {\cal R}^2_{i\beta}
 \end{eqnarray}
where $i=1,2,3$ and $\beta=0,1,2,3$. Explicitly, we have
%elements $ \omega_{ij}$ write
%$$ \omega_{ij} = \delta_{ij} \bigg[ \frac{1}{4}\sum_{\beta}\bigg( {\cal R}^2_{0\beta} - \sum_k {\cal R}^2_{k\beta} \bigg) \bigg]
%+ \frac{1}{2}\sum_{\beta} {\cal R}_{i\beta}{\cal R}_{j\beta}$$
%where $\beta=0,1,2,3$ and $k=1,2,3$. The diagonal elements are
%$$ \omega_{ii} = \frac{1}{4} \bigg[ \sum_{\beta}\bigg( {\cal R}^2_{0\beta} - \sum_k {\cal R}^2_{k\beta} \bigg) \bigg]
%+ \frac{1}{2}\sum_{\beta} {\cal R}^2_{i\beta}$$
%and the off-diagonal elements are
%$$ \omega_{ij} = \frac{1}{2}\sum_{\beta} {\cal R}_{i\beta} {\cal R}_{j\beta} \quad i\neq j$$
%Explicitly, we have
\begin{eqnarray} \label{omega11}
 \omega_{11}=  \frac{1}{4}\bigg[ 4 \bigg(\sqrt{\lambda_1} + \sqrt{\lambda_4} \bigg)\bigg(\sqrt{\lambda_2} + \sqrt{\lambda_3} \bigg) +
\frac{(R^2_{11}-R^2_{22})+(R^2_{03}-R^2_{30})}{(\sqrt{\lambda_1} +
\sqrt{\lambda_4} )(\sqrt{\lambda_2} + \sqrt{\lambda_3} )} \bigg]
 \end{eqnarray}
%\begin{eqnarray} \label{omega11}
% \omega_{11}=  \frac{1}{4}\bigg[ 4 \bigg(\sqrt{\lambda_1} + \sqrt{\lambda_4} \bigg)\bigg(\sqrt{\lambda_2} + \sqrt{\lambda_3} \bigg) +
%\frac{(R^2_{11}-R^2_{22})+(R^2_{12}-R^2_{21})+(R^2_{03}-R^2_{30})}{(\sqrt{\lambda_1}
%+ \sqrt{\lambda_4} )(\sqrt{\lambda_2} + \sqrt{\lambda_3} )} \bigg]
% \end{eqnarray}
\begin{eqnarray}\label{omega22}
\omega_{22}=  \frac{1}{4}\bigg[ 4 \bigg(\sqrt{\lambda_1} +
\sqrt{\lambda_4} \bigg)\bigg(\sqrt{\lambda_2} + \sqrt{\lambda_3}
\bigg) +
\frac{(R^2_{22}-R^2_{11})+(R^2_{03}-R^2_{30})}{(\sqrt{\lambda_1} +
\sqrt{\lambda_4} )(\sqrt{\lambda_2} + \sqrt{\lambda_3} )} \bigg]
 \end{eqnarray}
%\begin{eqnarray}\label{omega22}
%\omega_{22}=  \frac{1}{4}\bigg[ 4 \bigg(\sqrt{\lambda_1} +
%\sqrt{\lambda_4} \bigg)\bigg(\sqrt{\lambda_2} + \sqrt{\lambda_3}
%\bigg) +
%\frac{(R^2_{22}-R^2_{11})+(R^2_{21}-R^2_{12})+(R^2_{03}-R^2_{30})}{(\sqrt{\lambda_1}
%+ \sqrt{\lambda_4} )(\sqrt{\lambda_2} + \sqrt{\lambda_3} )} \bigg]
% \end{eqnarray}
\begin{eqnarray}\label{omega33}
\omega_{33} &=& \frac{1}{2}\bigg[  \bigg(\sqrt{\lambda_1} +
\sqrt{\lambda_4} \bigg)^2 + \bigg(\sqrt{\lambda_2} +
\sqrt{\lambda_3} \bigg)^2 \bigg]
 + \frac{1}{8}\bigg[
\frac{(R_{03}+R_{30})^2-(R_{11}-R_{22})^2}{\bigg(\sqrt{\lambda_1} +
\sqrt{\lambda_4} \bigg)^2}\bigg]\nonumber
\\ && +\frac{1}{8}\bigg[
\frac{(R_{03}-R_{30})^2-(R_{11}+R_{22})^2}{\bigg(\sqrt{\lambda_2} +
\sqrt{\lambda_3} \bigg)^2 }\bigg]
 \end{eqnarray}
%\begin{eqnarray}\label{Omega33}
%\omega_{33} &=& \frac{1}{2}\bigg[  \bigg(\sqrt{\lambda_1} +
%\sqrt{\lambda_4} \bigg)^2 + \bigg(\sqrt{\lambda_2} +
%\sqrt{\lambda_3} \bigg)^2 \bigg]
% + \frac{1}{8}\bigg[
%\frac{(R_{03}+R_{30})^2-(R_{11}-R_{22})^2-(R_{12}+R_{21})^2}{\bigg(\sqrt{\lambda_1}
%+ \sqrt{\lambda_4} \bigg)^2}\bigg]\nonumber
%\\ && +\frac{1}{8}\bigg[
%\frac{(R_{03}-R_{30})^2-(R_{11}+R_{22})^2-(R_{12}-R_{21})^2}{\bigg(\sqrt{\lambda_2}
%+ \sqrt{\lambda_3} \bigg)^2 }\bigg]
% \end{eqnarray}
%\begin{eqnarray}\label{Omega12}
% \omega_{12} =  \omega_{21} =  \frac{1}{2}\bigg({\cal R}_{11} {\cal R}_{21}+ {\cal R}_{12} {\cal R}_{22} \bigg)=
%\frac{1}{2} \frac{R_{11}R_{21}+ R_{22}R_{12}}{(\sqrt{\lambda_1} +
%\sqrt{\lambda_4})(\sqrt{\lambda_2} + \sqrt{\lambda_3})}
%\end{eqnarray}
%\begin{eqnarray}\label{Omega13}
%\omega_{13} =  \omega_{31} = 0 ,\quad \omega_{23} =  \omega_{32} = 0
%\end{eqnarray}


Clearly, the above results were derived for $X$ states satisfying
$t_1\neq 0$ and $t_2\neq 0$. The situation $t_1 = 0$ occurs when
$\rho_{11} = \rho_{44} = 0$ and the positivity condition of  $\rho$ (\ref{eqn1}) implies $ \rho_{14} =
0$. In this case the correlation matrix elements of the matrix $\sqrt{\rho}$ write simply as

$$  {\cal R}_{00} =   \sqrt{t_2 + 2 \sqrt{d_2}}   \qquad  {\cal R}_{03} =   -  \frac{1}{2} \frac{R_{30}- R_{03} }{\sqrt{t_2 + 2\sqrt{d_2}}}$$
$$  {\cal R}_{30} =  \frac{1}{2} \frac{R_{30}- R_{03} }{\sqrt{t_2 + 2\sqrt{d_2}}} \qquad {\cal R}_{11} = \frac{1}{2} \frac{R_{11}  + R_{22}  }{\sqrt{t_2 + 2\sqrt{d_2}}} $$
%$$  {\cal R}_{12} =  \frac{1}{2} \frac{R_{12}  + R_{21}  }{\sqrt{t_1 + 2\sqrt{d_1}}} +  \frac{1}{2} \frac{R_{12}- R_{21} }{\sqrt{t_2 + 2\sqrt{d_2}}} \qquad  {\cal R}_{21} = \frac{1}{2} \frac{R_{12}  + R_{21}  }{\sqrt{t_1 + 2\sqrt{d_1}}} -  \frac{1}{2} \frac{R_{12}- R_{21} }{\sqrt{t_2 + 2\sqrt{d_2}}}$$
$$  {\cal R}_{22} = \frac{1}{2} \frac{R_{11}  + R_{22}  }{\sqrt{t_2 + 2\sqrt{d_2}}}  \qquad  {\cal R}_{33} =   -  \sqrt{t_2 + 2 \sqrt{d_2}} $$



Similarly, in the special case where $t_2= 0$ (or equivalently $\rho_{22} = \rho_{33} = \rho_{23} = \rho_{32} = 0$), one has
$$  {\cal R}_{00} =  \sqrt{t_1 + 2 \sqrt{d_1}}    \qquad  {\cal R}_{03} =   \frac{1}{2} \frac{R_{30}  + R_{03}  }{\sqrt{t_1 + 2\sqrt{d_1}}} $$
$$  {\cal R}_{30} = \frac{1}{2} \frac{R_{30}  + R_{03}  }{\sqrt{t_1 + 2\sqrt{d_1}}}  \qquad {\cal R}_{11} =  \frac{1}{2} \frac{R_{11}- R_{22} }{\sqrt{t_1 + 2\sqrt{d_1}}}$$
%$$  {\cal R}_{12} =  \frac{1}{2} \frac{R_{12}  + R_{21}  }{\sqrt{t_1 + 2\sqrt{d_1}}} +  \frac{1}{2} \frac{R_{12}- R_{21} }{\sqrt{t_2 + 2\sqrt{d_2}}} \qquad  {\cal R}_{21} = \frac{1}{2} \frac{R_{12}  + R_{21}  }{\sqrt{t_1 + 2\sqrt{d_1}}} -  \frac{1}{2} \frac{R_{12}- R_{21} }{\sqrt{t_2 + 2\sqrt{d_2}}}$$
$$  {\cal R}_{22} =  -  \frac{1}{2} \frac{R_{11}- R_{22} }{\sqrt{t_1 + 2\sqrt{d_1}}} \qquad  {\cal R}_{33} =  \sqrt{t_1 + 2 \sqrt{d_1}}   $$


For $t_1= 0$, the results (\ref{omega11}), (\ref{omega22}) and (\ref{omega33}) give

\begin{equation}
\omega_{11} = 0 \quad \omega_{22} = 0  \quad \omega_{33} = \frac{1}{2}\bigg[    t_2 + 2 \sqrt{d_2} \bigg]
 + \frac{1}{8}\bigg[
\frac{(R_{03}-R_{30})^2-(R_{11}+R_{22})^2}{ t_2 + 2 \sqrt{d_2}}\bigg]
 \end{equation}


The quantity $\omega_{33}$ can be written also as

\begin{equation}
 \omega_{33} = \frac{1}{2}\bigg[    1  + 2 \sqrt{\rho_{22}\rho_{33}-{\rho^2_{23}}} \bigg]
 + \frac{1}{2} \bigg[  \frac{(\rho_{22}\rho_{33})^2- 2\rho_{23}^2}{1  + 2 \sqrt{\rho_{22}\rho_{33}-\rho_{23}^2}}\bigg]
 \end{equation}


Similarly,  for $t_2= 0$ one gets


\begin{equation}
\omega_{11} = 0 \quad \omega_{22} = 0  \quad \omega_{33} = \frac{1}{2}\bigg[    t_1 + 2 \sqrt{d_1} \bigg]
 + \frac{1}{8}\bigg[
\frac{(R_{03}+R_{30})^2-(R_{11}-R_{22})^2}{ t_1 + 2 \sqrt{d_1}}\bigg]
 \end{equation}

In this special case, the quantity $\omega_{33}$ can be explicitly given by
\begin{equation}
 \omega_{33} = \frac{1}{2}\bigg[    1  + 2 \sqrt{\rho_{11}\rho_{44}-{\rho^2_{14}}} \bigg]
 + \frac{1}{2} \bigg[  \frac{(\rho_{11}\rho_{44})^2- 2\rho_{14}^2}{1  + 2 \sqrt{\rho_{11}\rho_{44}-\rho_{14}^2}}\bigg]
 \end{equation}
in terms of the non vanishing density matrix elements.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{Special two qubit $X$ states}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsubsection{Two qubit Bell states }
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

The two qubit $X$ states (\ref{fano-bloch-rho}) becomes of Bell type when the Bloch correlations elements $R_{30}$ and $R_{03}$ are zero. In this case, using the
results (\ref{omega11}), (\ref{omega22}) and (\ref{omega33}), the eigenvalues of the matrix $ \omega$ (cf. equation (\ref{matrixomega})) write
\begin{equation}\label{omegabell11}
\omega_{11} = 2 \bigg(\sqrt{\lambda_1 \lambda_2} +
\sqrt{\lambda_3\lambda_4 } \bigg)
\end{equation}
\begin{equation}\label{omegabell22}
\omega_{22} = 2 \bigg(\sqrt{\lambda_1 \lambda_3} +
\sqrt{\lambda_2\lambda_4 } \bigg)
\end{equation}
\begin{equation}\label{omegabell33}
\omega_{33} = 2 \bigg(\sqrt{\lambda_1 \lambda_4} +
\sqrt{\lambda_2\lambda_3 } \bigg)
\end{equation}
which coincide with  the results derived in \cite{Karimipour}. The eigenvalues
$\omega_{11}$, $\omega_{22}$ and $\omega_{33}$ rewrites also as
\begin{equation}\label{omegabell111}
\omega_{11} = \frac{1}{2} \bigg(\sqrt{(1- R_{11})^2 - (R_{22} +
R_{33})^2} + \sqrt{(1+ R_{11})^2 - (R_{22} - R_{33})^2 } \bigg)
\end{equation}
\begin{equation}\label{omegabell222}
\omega_{22} = \frac{1}{2} \bigg(\sqrt{(1- R_{22})^2 - (R_{33} +
R_{11})^2} + \sqrt{(1+ R_{22})^2 - (R_{33} - R_{11})^2 } \bigg)
\end{equation}
\begin{equation}\label{omegabell333}
\omega_{33} = \frac{1}{2}\bigg(\sqrt{(1- R_{33})^2 - (R_{11} +
R_{22})^2} + \sqrt{(1+ R_{33})^2 - (R_{11} - R_{22})^2 } \bigg)
\end{equation}
in terms of the correlation elements $R_{11}$, $R_{22}$ and $R_{33}$


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsubsection{Orthogonal invariant two-qubit states}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

The local quantum uncertainty for orthogonal invariant two qubit states was  discussed in \cite{Sen}.
Any two qubit state invariant under the operation ${\cal
O}\otimes{\cal O}$ ( with ${\cal O}$  an arbitrary orthogonal matrix) can be
expanded in terms of the three generators $\mathcal{I}$ ,
$\mathcal{F}_1$ and $\mathcal{F}_2$ as \cite{Sen}
\begin{equation}\label{O-invariant}
\rho = a \mathcal{I} + b \mathcal{F}_1 + c \mathcal{F}_2
\end{equation}
where the real parameters $a$, $b$ and $c$ are positive and  satisfy
$4a+2b+2c=1$ (trace condition), $\mathcal{I}$ is the identity and
the operators $\mathcal{F}_1$ and   $\mathcal{F}_2$
$$\mathcal{F}_1 = \sum_{ij} \vert ij \rangle \langle ji \vert \quad  \mathcal{F}_2 = \sum_{ij}  \vert ii \rangle \langle jj \vert$$
in the computational basis. The density matrix (\ref{O-invariant}) has the form $X$
\begin{equation}
\rho =  \left(
\begin{array}{cccc}
a+b+c & 0 & 0 & c \\
0 & a & b & 0 \\
0 & b & a & 0 \\
c & 0 & 0 & a+b+c
\end{array}
\right).
\end{equation}
and using the results (\ref{omega11}), (\ref{omega22}) and (\ref{omega33}), one verifies
$$ \omega_{11} = \omega_{33} =  2 \bigg(\sqrt{(a+b)(a+b+2c)} + \sqrt{a^2-b^2 } \bigg) \quad \omega_{22} =  2 \bigg(\sqrt{(a-b)(a+b+2c)} + (a+b) \bigg)$$
and one recovers the results obtained in \cite{Sen}.























%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%\subsection{Quantum correlation in separable states}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



%In this situation the trace quantum discord is
%\begin{eqnarray}
% D_{\rm g}(\rho)= \frac{1}{2}(1 - \alpha) + \sqrt{c_1(\alpha - c_1)}.
%\end{eqnarray}



%For $ \frac{2}{3} \leq \alpha \leq 1 $,


%we have $ \vert R_{11}\vert
%\leq \vert R_{33}\vert $ when $ c_1 \in [ \alpha_- , \alpha_+] $ and
%$\vert R_{33}\vert \leq \vert R_{11}\vert $ when $c_1 \in [
%0,\alpha_-] \cup [ \alpha_+, \alpha ]$ where
%$$\alpha_{\pm} = \frac{\alpha}{2} \pm \sqrt{\frac{5}{2}\alpha^2-3\alpha +1}.$$
%Accordingly, for $ \frac{2}{3} \leq \alpha \leq 1 $, the trace discord is given by
%\begin{eqnarray}
% D_{\rm g}(\rho)= \frac{1}{2}(1 - \alpha) + \sqrt{c_1(\alpha - c_1)}.
%\end{eqnarray}
%for $ c_1 \in [ \alpha_- , \alpha_+] $ and
%\begin{eqnarray}
% D_{\rm g}(\rho)= \frac{1}{2}\bigg( 1 - (\sqrt{c_1} + \sqrt{\alpha - c_1})^2\bigg)^2 + 4 \sqrt{c_1(\alpha - c_1)}(\sqrt{c_1} - \sqrt{\alpha - c_1})^2
%\end{eqnarray}
%when $c_1 \in [
%0,\alpha_-] \cup [ \alpha_+, \alpha ]$.




%we have $ \vert R_{33}\vert
%\leq \vert R_{11}\vert $ when $ c_1 \in [ \alpha_- , \alpha_+] $ and
%$\vert R_{11}\vert \leq \vert R_{33}\vert $ when $c_1 \in [
%0,\alpha_-] \cup [ \alpha_+, \alpha ]$ where
%$$\alpha_{\pm} = \frac{\alpha}{2} \pm (1-\alpha)(2\alpha -1).$$

%Accordingly, the trace discord is given by
%\begin{eqnarray}
% D_{\rm g}(\rho)= \frac{1}{2}(1 - \alpha) + \sqrt{c_1(\alpha - c_1)}.
%\end{eqnarray}
%for $ c_1 \in [ \alpha_- , \alpha_+] $
%and
%\begin{eqnarray}
% D_{\rm g}(\rho)= \frac{1}{2}\bigg( 1 - (\sqrt{c_1} + \sqrt{\alpha - c_1})^2\bigg)^2 + 4 \sqrt{c_1(\alpha - c_1)}(\sqrt{c_1} - \sqrt{\alpha - c_1})^2
%\end{eqnarray}
%when $c_1 \in [
%0,\alpha_-] \cup [ \alpha_+, \alpha ]$





%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{local quantum uncertainty in symmetric multi-qubit
systems}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


The multi-qubit symmetric states were shown relevant  for different
purposes  in quantum information science~\cite{solano,
mixed,usa1,usa2,markham1,markham2,gebastin,markham3}. In this paper,
we shall mainly focus on an ensemble of $n$ spin-$1/2$ prepared in
even and odd spin coherent states.

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{Symmetric multi-qubit systems}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

We consider $n$ identical qubits. Each qubit lives in a
2-dimensional Hilbert space
$\mathcal{H}=\mathrm{span}\{\vert{0}\rangle,\vert{1}\rangle\}$. The
Hilbert space of the $n$-qubit system is given  by $n$ tensored
copies of $\mathcal{H}$
$$\mathcal{H}_n := \mathcal{H}^{\otimes n}.$$
Among the multi-partite states in $\mathcal{H}_n$, multi-qubit
states obeying exchange symmetry are of special interest from
experimental as well as mathematical point of views. An arbitrary
symmetric $n$-qubit state is commonly represented in either
Majorana~\cite{Maj32} or  Dicke ~\cite{Dic54} representation. Any
multi-qubit state, invariant under the exchange symmetry, is
specified in the Majorana description by the state (up to a
normalization factor)
\begin{equation}\label{psisym}
|\psi_s\rangle = \frac{1}{n!} \sum_{\sigma \in {\cal S}_n} \vert
{\eta_{\sigma(1)},\ldots,\eta_{\sigma(n)}}\rangle,
\end{equation}
where each single qubit state is $|\eta_i\rangle \equiv (1 + \eta_i
\bar\eta_i)^{-\frac{1}{2}}(|0\rangle + \eta_i |1\rangle)~
(i=1,\ldots, n)$
%$|\eta_i\rangle \equiv \alpha_i |0\rangle + \beta_i |1\rangle$ ($i=1,\ldots, n$)
 and
the sum is over the elements of the permutation group ${\cal S}_n$
of $n$ objects. In Equation (\ref{psisym}), the vector
$\vert{\eta_{\sigma(1)},\ldots,\eta_{\sigma(n)}}\rangle$ stands for
the tensor product
$\vert{\eta_{\sigma(1)}}\rangle\otimes\ldots\otimes\vert{\eta_{\sigma(n)}}\rangle$.
The totally symmetric $n$-qubit states can be also formulated  in
Dike representation. The symmetric Dicke states with $k$ excitations
are defined by ~\cite{Dic54}
%\begin{equation}
%\label{Dicke}
%\vert {D_n(k)} \rangle = \frac{1}{\sqrt{C_n^k}} \sum_{\sigma} \vert{\underbrace{0\ldots 0}_{n-k}\underbrace{1\ldots 1}_{k}}\rangle,
%\end{equation}
\begin{equation}
\label{Dicke} \vert n,k \rangle = \sqrt{\frac{k!(n-k)!}{n!}}
\sum_{\sigma \in {\cal S}_n} \vert{\underbrace{0,\ldots,
0}_{n-k},\underbrace{1,\ldots, 1}_{k}}\rangle,
\end{equation}
which generate an orthonormal basis of the  symmetric Hilbert
subspace of dimension $(n+1)$. Therefore, permutation invariance, in
symmetric multi-qubit states, implies a  restriction to $n+1$
dimensional subspace from the entire  $2^n$ dimensional Hilbert
space. The Dicke states (\ref{Dicke}) constitute a special subset of
the symmetric multi-qubit states (\ref{psisym}) corresponding to the
situation where the first $k$-qubit are such that $\eta_i = 0$ for
$i = 0,1, \ldots k$ and the remaining qubits are in the states
$\vert \eta_i = 1 \rangle $ with $i = k+1, \ldots , n$. The states
(\ref{Dicke}) are the eigenstates of the collective spin operators
$J^2$ and $J_z$ defined as
$$ J_{\alpha} = \frac{1}{2} \sum_{i=1}^n \sigma_{i\alpha} \quad \alpha = x,y,z $$
where the operators $\sigma_{i\alpha}$ stand for the spin
$\frac{1}{2}$-Pauli operators. In this respect,  the symmetric qubit
states (\ref{Dicke}) are completely determined by the quantum
angular momentum $j = 2n$ which may take integer or half integer
values ( $j = {1\over 2}, 1, \frac{3}{2}, \ldots$) specifying  the
irreducible representations classes of the group $SU (2)$. The
$(2j+1)$-dimensional Hilbert space is spanned by the irreducible
tensorial set $\{ \vert j , m \rangle, m = -j , -j+1, \cdots, j-1,j
\} \equiv \{ \vert n , k \rangle = \vert j = \frac{n}{2}, m + j
\rangle, k = 0, 1, \cdots, n\}$ characterizing the spin-$j$
representations of the group $SU(2)$.\\







\noindent As we shall consider the pairwise local quantum uncertainty in
symmetric states of type (\ref{psisymD}), we need the reduced two-qubit
density matrices extracted from the whole $n$ particles system. The general form of bipartite density matrix writes
as
\begin{equation}
\rho _{ij}=\left(
\begin{array}{llll}
\rho_{11} & \rho_{21}^{*} & \rho_{31}^{*} & \rho_{41}^{*} \\
\rho_{21} & \rho_{22} & \rho_{32}^{*} & \rho_{42}^{*} \\
\rho_{31} & \rho_{32} & \rho_{33} & \rho_{43}^{*} \\
\rho_{41} & \rho_{42} & \rho_{43} & \rho_{44}
\end{array}
\right)   \label{eq:rhogood}
\end{equation}
in the computational basis $\{|00\rangle ,|01\rangle ,|10\rangle
,|11\rangle \}$ where the matrix elements are given by
\begin{equation}
\rho_{11} =\frac 14\left( 1+ 2\langle \sigma _{i3} \otimes \sigma
_{j0} \rangle +\langle \sigma _{i3}\otimes \sigma _{j3}\rangle
\right) \qquad \rho_{44} =\frac 14\left( 1- 2\langle \sigma
_{i3}\otimes  \sigma _{j0}\rangle +\langle \sigma _{i3} \otimes
\sigma _{j3}\rangle \right)
\end{equation}
\begin{equation}
\rho_{21} = \rho_{31} = \frac 12(\langle \sigma _{i+} \otimes \sigma
_{j0}\rangle + \langle \sigma _{i+} \otimes \sigma _{j3}\rangle )
\qquad \rho_{42} = \rho_{43} = \frac 12(\langle \sigma _{i+} \otimes
\sigma _{j0}\rangle -\langle \sigma _{i+} \otimes \sigma
_{j3}\rangle )
\end{equation}
\begin{equation}
\rho_{22} = \rho_{33}=\frac 14\left( 1-\langle \sigma _{i3} \sigma
_{i+}\otimes \sigma _{j3}\rangle \right)
\end{equation}
\begin{equation}
\rho_{32}  = \langle\sigma_{i+} \otimes \sigma_{j-}\rangle
\end{equation}
\begin{equation}
\rho_{41} =\frac 14(\langle \sigma _{i1} \otimes \sigma _{j1}\rangle
-\langle \sigma _{i2} \otimes \sigma _{j2}\rangle +i2\langle \sigma
_{i1}\otimes \sigma _{j2}\rangle ). \label{eq:para1}
\end{equation}
For  collective spin models, the pairwise reduced density matrix
in the standard basis, $\{|{\downarrow}{\downarrow}{\rangle},
|{\downarrow}{\uparrow}{\rangle},|{\uparrow}{\downarrow}{\rangle},
|{\uparrow}{\uparrow}{\rangle}\}$
(with ${\sigma_z}|{\uparrow}{\rangle}=|{%
\uparrow}{\rangle}$ and ${\sigma_z}|{\downarrow}{\rangle}=-|{\downarrow}{%
\rangle}$)~\cite{wang1}, can be derived in terms of the collective
operators spin. Indeed, for  states, with symmetry exchange, we have
$$  \langle \sigma_{i\alpha} \otimes \sigma_{j0}\rangle = \frac{\langle J_{\alpha}\rangle}{n}, \qquad \langle \sigma_{i\alpha} \otimes \sigma_{j\alpha}\rangle = \frac{4 \langle J^2_{\alpha}\rangle- n}{n(n-1)} $$
$$  \langle \sigma_{i1} \otimes \sigma_{j2}\rangle = \frac{\langle J_{1} J_{2} + J_{2}J_{1}\rangle}{n(n-1)}, \qquad \langle \sigma_{i+} \otimes \sigma_{j3}\rangle = \frac{4 \langle J_{+} J_{3} + J_{3}J_{+}\rangle}{n(n-1)} $$
where  $\alpha = 1, 2, 3$. It follows that the explicit expressions
for the elements of the reduced density matrix  are given by
\begin{eqnarray}
\rho_{11} =\frac{n^2-2n+4{\langle}J^{2}_{z}{\rangle}+4(n-1){\langle}
J_{z}{\rangle}}{4n(n-1)}, \quad \rho_{44}
=\frac{n^2-2n+4{\langle}J^{2}_{z}{\rangle}-4(n-1){\langle}
J_{z}{\rangle}}{4n(n-1)}
\end{eqnarray}
\begin{eqnarray}
\rho_{21} = \rho_{31}
=\frac{(n-1){\langle}J_{+}{\rangle}+{\langle}J_{+}J_{z}+ J_{z}J_{+}{
\rangle}}{2n(n-1)},\quad \rho_{42} = \rho_{43}
=\frac{(n-1){\langle}J_{+}{\rangle}{\pm}{\langle}J_{+}J_{z}+
J_{z}J_{+}{ \rangle}}{2n(n-1)}
\end{eqnarray}
\begin{eqnarray}
\rho_{22} = \rho_{33}= \rho_{23} = \rho_{32}=
\frac{n^2-4{\langle}J^2_{z}{\rangle}}{4n(n-1)}=\frac{{\langle}
J^2_x+J^2_y{\rangle}-n/2}{n(n-1)},
\end{eqnarray}
\begin{eqnarray}
\rho_{41} = \frac{{\langle}J^2_+{\rangle}}{n(n-1)},
\end{eqnarray}
%$w=y$, for $\sum_{\alpha=x,y,z}J^2_{\alpha}=J^2=
%\frac{N}{2}(\frac{N}{2}+1)$.
%For the symmetric states with parity conservation, we find
%$x_{\pm}=0$
For  states with parity symmetry,  the
density matrix commutes with the operator $\sigma_3 \otimes
\sigma_3$. This implies $\rho_{12} = \rho_{13} = \rho_{42} =
\rho_{43}= 0$. In fact, for states with parity symmetry, we have $
\langle J_{1}\rangle = \langle J_{2} \rangle = 0$ and $ \langle
J_{1}J_{3}\rangle = \langle J_{2}J_{3} \rangle = 0$. Hence the
pairwise reduced density matrix is $X$ shaped  and  writes as

\begin{equation}
\rho _{ij}=\left(
\begin{array}{llll}
\rho_{11} & 0 & 0 & \rho_{41}^{*} \\
0 & \rho_{22} & \rho_{22} & 0 \\
0 & \rho_{22} & \rho_{22} & 0 \\
\rho_{41} & 0 & 0 & \rho_{44}
\end{array}
\right)   \label{rho:2}
\end{equation}




We note that the local unitary transformation
$$ \vert 0 \rangle_k \rightarrow \exp \bigg(\frac{i}{2} (\theta)\bigg) \vert 0 \rangle_k $$
eliminates the phase factors of the matrix element $\rho_{41}$ with
$\rho_{41} = \vert\rho_{41} \vert e^{i\theta}$ and $k=1,2$ labels
the subsystems 1 and 2. It follows that non-zero elements of the
correlation matrix take the simple form
\begin{equation}\label{matrix-R-special}
R_{03} = R_{30} = \rho_{11} - \rho_{44} \quad R_{11} = 2( \rho_{22} + \vert \rho_{41}\vert) \quad R_{22} = 2( \rho_{22} - \vert \rho_{41}\vert) \quad R_{33} = 1 - 4 \rho_{22}
\end{equation}



\subsection{Pairwise Local quantum uncertainty in Dicke
states}


The Dicke states are defined as the equal superposition of all basis
states of $n$ qubits having exactly $k$ excitations (\ref{Dicke}). Nowadays it is
commonly accepted that this family of symmetric states can be an
useful resource in various quantum protocols for two main reasons.
First, they can be generated experimentally. indeed, the generation
of Dicke states with trapped-ion qubits have been proposed \cite{Retzker}. On the other hand,  quantum correlations  in Dicke states
are highly robust in presence of  external decoherence effects and
especially measurements on individual qubits \cite{Stockton}.  From the equation (\ref{eq:bbbb}), it is simply verified that the
reduced density matrix $\rho _{12}$, describing  two qubits extracted from the state (\ref{Dicke}), is given by
\begin{equation}
\rho _{12}=\left(
\begin{array}{llll}
\rho_{11} & 0 & 0 & \rho_{14}\\
0 & \rho_{22} & \rho_{23} & 0 \\
0 & \rho_{32} & \rho_{33} & 0 \\
\rho_{41} & 0 & 0 & \rho_{44}
\end{array}
\right)   \label{eq:rho}
\end{equation}
with $\rho_{14}= \rho_{41}=0$ and the non vanishing elements are
\begin{eqnarray}
\rho_{11} &=&\frac{k(k-1))}{n(n-1)},  \nonumber \\
\rho_{44} &=&\frac{(n-k)(n-k-1)}{n(n-1)},  \nonumber \\
 \rho_{22} &=& \rho_{23} ~=~ \rho_{32}~ = ~\rho_{33} ~=~ \frac{k(n-k)}{n(n-1)}.  \label{eq:elements}
\end{eqnarray}
From the correlation matrix elements (\ref{matrix-R-special}), one obtains
\begin{eqnarray}
R_{11} &=& R_{22} ~=~  \frac{2k(n-k))}{n(n-1)},  \nonumber \\
R_{33} &=& 1 - \frac{4k(n-k)}{n(n-1)},  \nonumber \\
R_{03} &=& R_{30} ~=~ \frac{2k-n}{n}.  \label{eq:elements}
\end{eqnarray}
Using the expressions (\ref{omega11}), (\ref{omega22}) and
(\ref{omega33}), one finds
\begin{equation}
\omega_{11} = \omega_{22} = \sqrt{\frac{2k(n-k)}{n(n-1)}} ~ \bigg(
\sqrt{k(k-1)} + \sqrt{(n-k)(n-k-1)}\bigg)
\end{equation}
\begin{equation}
\omega_{33} = 1 + 2 ~\frac{k(k-n)}{n(n-1)}
\end{equation}



%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\subsection{Local quantum uncertainty in spin coherent states}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%


Any symmetric state  $|\psi_s \rangle$ (\ref{psisym}) can be
expanded in terms of Dicke states (\ref{Dicke}) as follows
\begin{equation}
\label{psisymD} |\psi_s\rangle = \frac{1}{n!} \sum_{k=0}^n c_k ~|n,k
\rangle,
\end{equation}
where  the $c_k$ ($k=0,\ldots,n$) stand for the complex expansion
coefficients. In particular, when the qubit are all identical
($\eta_i = \eta$ for all qubits), it is simply verified that the
coefficients $c_k$ are given by
\begin{equation}
\label{ck} c_k = n!  \sqrt{\frac{n!}{k!(n-k)!}} \frac{\eta^{k}}{(1 +
\eta\bar\eta)^{\frac{n}{2}}}
\end{equation}
and the symmetric multi-qubit states  (\ref{psisym}) write
\begin{equation}\label{psisym-cs}
|\psi_s\rangle :=  \vert n, \eta \rangle = (1 +
\eta\bar\eta)^{-\frac{n}{2}} \sum_{k=0}^n \sqrt{\frac{n!}{k!(n-k)!}}
\eta^{k}  ~|n,k \rangle,
\end{equation}
which are exactly the  $j=\frac{n}{2}$-spin coherent states (see for instance \cite{daoud2}). In particular, the
state $\vert n, \eta \rangle $ can be identified for $n=1$ with
spin-$\frac{1}{2}$ coherent state with $|0\rangle \equiv
|\frac{1}{2}, -\frac{1}{2}\rangle$ and $|1\rangle \equiv
|\frac{1}{2} ,+ \frac{1}{2}\rangle$). The standard $ SU(2)$ coherent
states are obtained by the action of an element of the coset space
$SU(2)/ U(1)$
\begin{equation}
 D_j(\xi) = \exp (\xi J_{+} - \xi^{\ast} J_{-}) \ ,
\end{equation}
on the extremal state $|j,-j\rangle$. This action  gives  the states
\begin{equation}
|j,\eta\rangle = D_j(\xi) |j,-j\rangle = \exp(\xi J_{+} - \xi^{\ast}
J_{-}) |j,-j\rangle = (1+|\eta|^{2})^{-j} \exp(\eta J_{+})
|j,-j\rangle \ ,   \label{4.7}
\end{equation}
where $\eta = (\xi/|\xi|)\tan |\xi|$. In the standard angular momentum basis $\{ \vert j , m
\rangle\}$, they write
\begin{equation}\label{cs-spinj}
|j,\eta\rangle = (1+|\eta|^{2})^{-j} \sum_{m=-j}^{j} \left[
\frac{(2j)!}{(j+m)!(j-m)!} \right]^{1/2} \eta^{j+m} |j,m\rangle \ .
\end{equation}
They satisfy the resolution to identity property
\begin{equation}
\int d\mu(j,\eta) |j,\eta\rangle \langle j,\eta| = I \ ,
\mbox{\hspace{1.0cm}} d\mu(j,\eta) = \frac{2j+1}{\pi}
\frac{d^{2}\!\eta}{(1+|\eta|^{2})^{2}} \ .  \label{4.10}
\end{equation}
The spin coherent states are not orthogonal to each other:
\begin{equation}
\langle j,\eta_{1}|j,\eta_{2}\rangle = (1+|\eta_{1}|^{2})^{-j}
(1+|\eta_{2}|^{2})^{-j} (1 + \eta_{1}^{\ast} \eta_{2})^{2j} \ .
\label{4.9}
\end{equation}
The resolution to identity makes possible to expand an arbitrary
state in terms of the coherent states $|j,\eta\rangle$. In the
special case $j = \frac{1}{2}$, the spin coherent states
(\ref{cs-spinj}) reduce to
\begin{equation}
|  \eta \rangle = \frac{1}{\sqrt{1 + \bar\eta \eta}}
|\downarrow\rangle
           + \frac{\eta}{\sqrt{1 + \bar\eta \eta}} |\uparrow\rangle . \label{coh}
\end{equation}
Here and in the following $| \eta \rangle$ is short for the
spin-$\frac{1}{2}$ coherent state $|{\frac{1}{2}}, \eta\rangle$ with
$|\uparrow\rangle \equiv |\frac{1}{2},\frac{1}{2}\rangle$ and
$|\downarrow \rangle \equiv |\frac{1}{2} ,- \frac{1}{2}\rangle$). It
is important to notice that the  tensorial product of two  $SU(2)$
coherent states $ |j_1, \eta \rangle$ and $ |j_2, \eta \rangle$
produces a spin-$(j_1+j_2)$ coherent state labeled by the same
variable:
\begin{equation}\label{split}
 |j_1, \eta \rangle \otimes | j_2 , \eta \rangle
  =  (D_{j_1} \otimes D_{j_2})\, (|j_1,j_1\rangle \otimes |j_2,j_2\rangle)
  =  D_{j_1+j_2}\, |j_1+j_2,j_1+j_2\rangle
    \; = \; |j_1 + j_2, \eta \rangle .
\end{equation}
Only coherent states possess this remarkable property. It allows  to
write any spin-$j$ coherent states as a $2j$ tensorial product of
spin-$\frac{1}{2}$
 coherent states:
\begin{equation}
|j, \eta \rangle =  \left(|\eta\rangle \right)^{\otimes 2j}
            =   \left( \frac{1}{\sqrt{1 + \bar\eta \eta}}
|\downarrow\rangle
           + \frac{\eta}{\sqrt{1 + \bar\eta \eta}} |\uparrow\rangle \right)^{\otimes 2j} \nonumber
          =  (1 + \bar\eta \eta)^{-j}\sum_{m=-j}^{+j} {2 j \choose j + m}^{\frac{1}{2}}
         \eta^{j+m}
          |j,m\rangle,  \label{Coh}
\end{equation}
reflecting that a spin-$j$ coherent state may be viewed as a
multipartite state containing  $2j$ qubits.

The even and odd spin coherent states are defined by
\begin{equation}\label{ncs}
 \vert j, \eta , m \rangle  =  {\cal N}_m ( \vert j, \eta \rangle + e^{im\pi} \vert j, - \eta
 \rangle)
\end{equation}
where the integer $m \in \mathbb{Z}$ takes the values $m = 0 ~({\rm
mod}~2)$ and $m = 1~ ({\rm mod}~2)$. The normalization factor ${\cal
N}_m$ is
$$ {\cal N}_m = \big[ 2 + 2 p^{2j} \cos m \pi\big]^{-1/2}$$
where  $p$ denotes the overlap between the states $\vert \eta
\rangle$ and $\vert  -\eta \rangle$. It is given by
\begin{equation}\label{overlap}
 p = \langle \eta \vert  - \eta \rangle = \frac{1 - \bar\eta \eta}{1 + \bar\eta \eta}.
\end{equation}
For  $ j = \frac{1}{2}$, the even and odd coherent states coincide
with $\vert \uparrow \rangle $ and $ \vert \downarrow \rangle$. They
can be identified with basis states for a logical qubit as $ \vert
0\rangle \rightarrow \vert \uparrow \rangle$ and $ \vert 1 \rangle
\rightarrow \vert \downarrow \rangle$. In this manner,  the states $
\vert j, \eta , m \rangle$ can be viewed as multipartite fermionic
coherent states:
\begin{equation}\label{cs-2j}
 \vert j, \eta , m \rangle  =  {\cal N}_m ( \left(|\eta \rangle \right)^{\otimes 2j} + e^{im\pi} \left(|-\eta \rangle \right)^{\otimes
 2j}).
\end{equation}



\noindent The decomposition property (\ref{split}) provides us with
a picture where even and odd spin coherent states can be considered
as comprising multipartite spin subsystems. This is our main
motivation to investigate the quantum correlations present in a
single spin coherent state. This issue is discussed in what follows.




\subsubsection{Pure bipartite spin coherent states}


Let us first  consider the following balanced superposition of spin coherent
states
\begin{equation}\label{cs-theta}
 \vert j, \eta , \theta \rangle  =  {\cal N}_{\theta} ( \vert j, \eta \rangle +
 e^{i\theta}  \vert j, -\eta \rangle)
\end{equation}
where the normalization factor is given by $|{\cal N}_{\theta}|^{-2}  = 2 + 2
p^{2j}\cos\theta $. Using the factorization or the splitting
property of spin coherent states (\ref{split}), the states
(\ref{cs-theta}) can be also expressed as
\begin{equation}\label{cs-2q}
 \vert j, \eta , \theta \rangle  =  {\cal N}_{\theta} ( \vert j_1, \eta \rangle\otimes\vert j_2, \eta \rangle +
 e^{i\theta} \vert j_1, -\eta \rangle\otimes\vert j_2, -\eta \rangle)
\end{equation}
with $j = j_1+j_2$. They can rewritten as a two qubit states in the
basis
$$ \vert
j_i, \eta , 0 \rangle \longrightarrow \vert  0 \rangle_{j_i} \qquad
\vert j_i, \eta , \pi\rangle \longrightarrow \vert  1 \rangle_{j_i},
\quad i=1,2.
$$
defined by means of  odd and even spin coherent associated with the
angular momenta $j_1$ and $j_2$. Indeed, for each subsystem, an
orthogonal basis $\{ \vert 0 \rangle_l , \vert 1 \rangle_l\}$, with
$ l = j_1$ or $j_2$, can be defined as
\begin{equation}\label{base0}
\vert 0 \rangle_l = \frac{ \vert l , \eta  \rangle +  \vert l ,
-\eta \rangle}{\sqrt{2(1 + p^{2l})}}
   \qquad \vert 1 \rangle_l = \frac{\vert l , \eta \rangle -  \vert l , -\eta
\rangle}{{\sqrt{2(1- p^{2l})}}}.
\end{equation}
The bipartite density state $\rho = \vert j, \eta , m\rangle \langle
j, \eta , m \vert$ is pure. The concurrence in this pure bipartite
system writes
\begin{equation}
{\cal C}_{j_1,j_2}(\theta) =
\frac{\sqrt{1-p^{4j_1}}\sqrt{1-p^{4j_2}}}{1+p^{2j}\cos
\theta}.\label{concurence1}
\end{equation}
Using the Schmidt decomposition, the state (\ref{cs-2q}) can be
written as
\begin{equation}\label{cs-shmidt}
 \vert j, \eta , \theta \rangle  =  \sqrt{\lambda_+} \vert + \rangle_1\otimes\vert + \rangle_2
 + \sqrt{\lambda_-}
 \vert - \rangle_1\otimes\vert - \rangle_2
\end{equation}
where $\lambda_{\pm}$ denote the eigenvalues of the reduced density
of the first subsystem $\rho_{j_1} = {\rm Tr}_{j_2} (\rho)$ obtained
by tracing out the spin $j_2$ . They write as
\begin{equation}\label{lambda}
\lambda_{\pm}= \frac{1}{2}\bigg( 1 \pm \sqrt{1 - {\cal C}^2} \bigg).
\end{equation}
in terms of the concurrence ${\cal C}\equiv{\cal
C}_{j_1,j_2}(\theta)$ given by (\ref{concurence1}). In the basis $\{
\vert + \rangle_1\otimes\vert + \rangle_2, \vert +
\rangle_1\otimes\vert - \rangle_2, \vert - \rangle_1\otimes\vert +
\rangle_2, \vert - \rangle_1\otimes\vert - \rangle_2 \}$, the
density matrix $ \rho_{j_1,j_2} (\theta) = \vert j, \eta , \theta
\rangle \langle j, \eta , \theta \vert $ takes the form
\begin{equation}
 \rho_{j_1,j_2} (\theta)=\left(
\begin{array}{llll}
\lambda_+ & 0 & 0 & \sqrt{\lambda_+\lambda_-}\\
0 & 0 & 0 & 0 \\
0 & 0 & 0 & 0 \\
\sqrt{\lambda_+\lambda_-} & 0 & 0 & \lambda_-
\end{array}
\right)   \label{eq:rhoj1j2}
\end{equation}
Using the result (\ref{omegaii}), one verifies that $\omega_{11} =
0$ , $\omega_{22} = 0$ and $\omega_{33} = 1 - 4\lambda_+\lambda_-$.
It follows that the local quantum uncertainty coincides with the squared
concurrence (\ref{concurence1})
\begin{equation}\label{LQU-pure}
 \mathcal{U}( \rho_{j_1,j_2} (\theta)) = {\cal C}^2_{j_1,j_2}(\theta).
\end{equation}
For $\theta = m \pi$ $(m \in \mathbb{Z})$, the logical qubits $
\vert j, \eta , m=0 \rangle$ and $ \vert j, \eta , m=1 \rangle$
coincide with even and odd spin coherent states. They behave like a
multipartite state of Greenberger-Horne-Zeilinger (${\rm GHZ}$) type
\cite{GHZ} in the asymptotic limit $p \rightarrow 0 $. Indeed, in
this limit, the  states $|\eta \rangle $ and $| -\eta \rangle $
approach orthogonality and an orthogonal basis can be defined such
that $\vert {\bf 0}\rangle\equiv \vert \eta \rangle$ and $\vert{\bf
1}\rangle \equiv \vert  -\eta \rangle$. Thus, the state $\vert j ,
\eta, m \rangle$ becomes of ${\rm GHZ}$-type
\begin{equation}
\vert j , \eta, m \rangle \sim \vert {\rm GHZ}\rangle_{2j} = \frac
1{\sqrt{2}}(\vert {\bf 0}\rangle \otimes |{\bf 0}\rangle \otimes
        \cdots \otimes\vert {\bf 0}\rangle
    +e^{i m \pi}\vert {\bf 1}\rangle \otimes
    \vert {\bf 1}\rangle \otimes \cdots \otimes
\vert {\bf 1}\rangle)\label{GHZ}
\end{equation}
which is maximally entangled and the bipartite local quantum
uncertainty is $\mathcal{U}( \rho_{j_1,j_2} (\theta=m\pi))= 1$.



Another interesting limiting case concerns the situation where $p^2
\rightarrow 1$ (or $ \eta \rightarrow 0$ ). In this case  the state $\vert j , \eta, m = 0 ~({\rm mod}~
2) \rangle$ (\ref{cs-2j}) reduces to ground state of a collection of
$2j$ fermions
\begin{equation}
\vert j,  0 , 0 ~({\rm mod}~ 2) \rangle \sim  \vert \downarrow
\rangle \otimes\vert \downarrow \rangle \otimes \cdots \otimes \vert
\downarrow \rangle,
\end{equation}
which is completely separable and
$$\mathcal{U}( \rho_{j_1,j_2} (\theta= m\pi)= 0.$$
The odd spin coherent state $\vert j , \eta , 1
~({\rm mod}~ 2) \rangle$  becomes a multipartite state of W
type~\cite{Dur00}
\begin{equation}
\vert j,  0 ,  1 ~({\rm mod}~ 2) \rangle \sim \vert\text{\rm
W}\rangle_{2j}
    = \frac{1}{\sqrt{2j}}(\vert \uparrow \rangle \otimes\vert \downarrow \rangle \otimes \cdots\otimes
       \vert \downarrow \rangle  +\vert \downarrow \rangle \otimes\vert \uparrow\rangle \otimes\ldots\otimes \vert \downarrow\rangle
      +\cdots
   + \vert  \downarrow \rangle \otimes\vert \downarrow \rangle  \otimes \cdots\otimes \vert \uparrow\rangle)~.
\label{Wstate}
\end{equation}
and, in this limiting situation, the local quantum uncertainty is
given by
$$ \mathcal{U}( \rho_{j_1,j_2} (\theta = \pi)) = 4 \frac{j_1j_2}{j_1+j_2}.$$
%The even spin coherent states  $\vert j, \eta , m = 0 ~({\rm mod}~2)
%\rangle$ interpolate continuously between ${\rm GHZ}_{2j}$ states
%$(p \rightarrow 0)$ and the completely separable state $\vert
%\downarrow \rangle \otimes\vert \downarrow \rangle \otimes \cdots
%\otimes \vert \downarrow \rangle$ $(p \rightarrow 1)$. In the odd
%case, corresponding  to
% $\vert j, \eta , m = 1 ~({\rm mod}~2) \rangle$, we obtain states
%interpolating between states of ${\rm GHZ}_{2j}$ type $(p
%\rightarrow 0)$ and states of Werner ${\rm W}_{2j}$ type $(p
%\rightarrow 1)$.
\subsubsection{Mixed bipartite states}
Now, we consider bipartite mixed density matrices  $\rho_{ij}$ obtained by a trace
procedure consisting in removing  all degrees of freedom of all qubits except the two qubits  $i$ and
$j$.  Since we consider quantum systems possessing the exchange symmetry, the trace procedure leads to  $2j(2j-1)/2$ identical density matrices $\rho_{12}$.
After some algebra, one gets
\begin{eqnarray}
\rho_{12} = {\cal N}^2(\vert \eta , \eta \rangle \langle \eta , \eta
\vert +\vert -\eta , -\eta \rangle \langle -\eta , -\eta | +
e^{im\pi } q |-\eta , -\eta \rangle \langle \eta , \eta \vert +e^{-i
m \pi }q \vert \eta , \eta \rangle \langle -\eta , -\eta \vert ).
\label{rhoij-vect}
\end{eqnarray}
The quantity $q$  occurring in (\ref{rhoij-vect}) is defined by
$$q = p^{2j-2}.$$

Setting $\eta = e^{i\phi}\sqrt{\frac{1-p}{1+p}}$, the density matrix
takes the form
\begin{equation}
\rho_{12} = \frac{1}{4(1+ p^{2j}\cos m\pi)} \left(
\begin{smallmatrix} (1+p)^2(1+q\cos m\pi)& 0
    & 0& e^{-2i\phi}(1-p^2)(1+q\cos m\pi
)\\
0  & (1-p^2)(1-q\cos m\pi ) & (1-p^2)(1-q\cos m\pi
) & 0 \\
0  & (1-p^2)(1-q\cos m\pi ) & (1-p^2)(1-q\cos m\pi
) & 0 \\
e^{2i\phi}(1-p^2)(1+q\cos m\pi ) & 0 & 0 & (1-p)^2(1+q\cos m\pi)
\end{smallmatrix}
\right) \label{rhoij}
\end{equation}
in the computational basis. The phase factor $\phi$ will be taken equal to zero. In other words,
the phase factor can be removed by a loacal transformation and the local quantum uncertainty remains unchanged as we discussed here above.
The bipartite mixed density $\rho_{12}$ (\ref{rhoij}) writes in
Fano-Bloch representation as
\begin{equation}
\rho_{12} = \sum_{\alpha \beta} R_{\alpha \beta}
\sigma_{\alpha}\otimes \sigma_{\beta}
\end{equation}
where the non vanishing matrix elements $R_{\alpha \beta}$ $(\alpha,
\beta = 0,1,2,3)$ are given by
$$ R_{00} = 1, \quad R_{11} =  \frac{1- p^2}{1+ p^{2j}\cos m\pi}, \quad R_{22} = \frac{(p^2-1)~p^{2j-2}\cos
m\pi}{1+ p^{2j}\cos m\pi},$$ $$ R_{33} = \frac{p^2 + p^{2j-2}\cos
m\pi}{1+ p^{2j}\cos m\pi}, \quad R_{03} =  R_{30} = \frac{p +
p^{2j-1}\cos m\pi}{1+ p^{2j}\cos m\pi}.$$



From the equations (\ref{omega11}), (\ref{omega22}) and
(\ref{omega33}), one obtains

\begin{equation}\label{om11}
\omega_{11} = \sqrt{\frac{1-p^2}{1+p^2}} ~\frac{\sqrt{1-
p^{4j-4}}}{1+ p^{2j}\cos m\pi}
\end{equation}
\begin{equation}\label{om22}
\omega_{22} =  p^2 \sqrt{\frac{1-p^2}{1+p^2}} ~\frac{\sqrt{1-
p^{4j-4}}}{1+ p^{2j}\cos m\pi}
\end{equation}
\begin{equation}\label{om33}
\omega_{33} = \frac{2p^2}{1+p^2} \frac{1+p^{2j-2}\cos m\pi}{1+
p^{2j}\cos m\pi}
\end{equation}

Since $ \omega_{22} \leq \omega_{11}$, we have $\omega_{\rm max} =
{\rm max}(\omega_{11},\omega_{22})$ and one verifies that

$${\rm sign} (\omega_{11}-\omega_{33}) = {\rm sign}\bigg(2(1-p^4)- (1+3p^4)(1+p^{2(j-1)}\cos m\pi)\bigg)$$

\textcolor[rgb]{1.00,0.00,0.00}{{\bf Etudier le signe dans les $j = 1, 3/2, 2, 5/2,  ....$ pour
$m=0$ et $m=1$}}

\textcolor[rgb]{1.00,0.00,0.00}{Tracer LQU en fonction de $p^2$ pour $j = 1, 3/2, 2, 5/2,  ....$}\\

We note that for the special case $j = 1$, the state
(\ref{cs-theta}) is a pure state with $j_1= j_2= 1/2$. In this case,
it is simple to verify that $\omega_{11} = \omega_{22} = 0$ and
$\omega_{33} = \frac{4p^2}{(1+p^2)^2}$ for $m=0$ and $\omega_{33} = 0 $
for $m=1$. Therefore, in this case the local quantum uncertainty writes
$$\mathcal{U}( \rho_{1/2,1/2}) = \frac{(1-p^2)^2}{(1+p^2)^2} $$ for $m=0$ and
$$\mathcal{U}( \rho_{1/2,1/2}) = 1 $$ for $m = 1$ in agreement with the result (\ref{LQU-pure}).

For $j> 1$ and $p \longrightarrow 0$, one obtains
$$ \omega_{11} = 1,  \omega_{22} = 0,  \omega_{33} = 0$$
and the local quantum uncertainty is zero.\\


Similarly, for $j> 1$ and $p^2 \longrightarrow 1$, it is simple to
check that the local quantum uncertainty vanishes for even spin
coherent states $(m=0)$. In this limiting situation, the matrix
elements (\ref{om11}), (\ref{om22}) and (\ref{om33}) become
$$ \omega_{11} = \frac{j-2}{\sqrt{2} j},  \omega_{22} = \frac{j-2}{\sqrt{2} j},  \omega_{33} = \frac{j-1}{ j}$$
for odd spin coherent states $(m=1)$. The local quantum uncertainty
is then given by
$$ \mathcal{U}( \rho_{12} (m=1)) \longrightarrow \frac{1}{j}.$$

\begin{center}
\includegraphics[width=4in]{fig1.eps}\\
{\bf Figure 1.}  {\sf The  local quantum uncertainty  
in symmetric states $(m=0)$ for different values of $j$.}
\end{center}
\begin{center}
\includegraphics[width=4in]{fig2.eps}\\
{\bf Figure 2.}  {\sf The local quantum uncertainty
in antisymmetric states $(m=1)$ for different values of $j$.}
\end{center}





%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\section{Concluding remarks}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%



In conclusion, we have derived the analytical expression of local quantum uncertainty for two-qubit
in $X$-states. This quantum correlations quantifier provides an efficient and computable way to characterize the nature
of correlations present in a multi-partite quantum system. Moreover, the analytical results obtained in this paper covers
some special class of two-qubit states recently investigated in the literature. We quote  Bell states and orthogonally invariant two-qubit
system examined in \cite{Karimipour} and  \cite{Sen}, respectively. We also evaluated the pairwise local quantum uncertainty in
multi-partite systems with exchange and parity symmetries.
As illustration, we quantified the pairwise quantum correlations in  balanced superpositions of Dicke states. Our interest in such
states is mainly motivated by their relevance in various collective spin systems such as Dicke model \cite{Dicke} and Lipkin-Meshkov-Glick \cite{LMG} model exhibiting
quantum phase transition.









\end{document}



\bibitem{ref6} L. Jakobczyk and A. Jamroz, Phys. Lett. A {\bf 333} (2004) 35 ;
M. Franca Santos, P. Milman, L. Davidovich and N. Zagury, Phys. Rev. A {\bf 73} (2006) 040305 ; M. Ikram, F. L. Li, and M. S. Zubairy, Phys. Rev. A {\bf 75}, 062336 (2007); A. Al-Qasimi and D. F. V. James, {\it ibid} {\bf 77}, 012117 (2007)

\bibitem{ref7} A. R. P. Rau, M. Ali and G. Alber, EPL {\bf 82}, 40002 (2008); X. Cao and H. Zheng, Phys. Rev. A {\bf 77} (2008) 022320 ; C. E. Lopez, G. Romero, F. Castra, E. Solano and J. C. Retamal, Phys. Rev. Lett. {\bf 101} (2008) 080503 ; M. Ali, G. Alber and A. R. P. Rau, J. Phys. B: At. Mol. Opt. Phys. {\bf 42} (2009) 025501


\bibitem{Rana} S. Rana and P. Parashar, Phys. Rev. A {\bf 85} (2012)  024102

\bibitem{Hassan} A.S.M. Hassan, B. Lari  and P. S. Joag, Phys. Rev. A {\bf 85} (2012) 024302.

\bibitem{Rau} S.Vinjanampathy and A.R.P Rau, J. Phys. A: Math. Theor. {\bf 45} (2012) 095303.

\bibitem{Zhou} J. Zhou and H. Guo, Phys. Rev. A {\bf 87} (2013) 062315.





\bibitem{Wineland} D. Leibfried, E. Knill, S. Seidelin, J. Britton, R.B. Blakestad,
J. Chiaverini, D.B. Hume, W.M. Itano, J.D. Jost, C. Langer, R.
Ozeri, R. Reichle and D.J. Wineland, Nature {\bf 438} (2005) 639.

\bibitem{Haroche} M. Brune, S. Haroche, J.M. Raimond, L. Davidovich and N. Zagury,
Phys. Rev. A {\bf 45} (1992) 5193



.

\bibitem{Fanchini2}  F.F. Fanchini, M.C. de Oliveira, L.K. Castelano  and M.F. Cornelio, Phys. Rev. A {\bf 87} (2013) 032317.

\bibitem{Modi} K. Modi, T. Paterek, W. Son, V. Vedral and M. Williamson
Phys. Rev. Lett. {\bf 104} ( 2010 ) 080501.

\bibitem{Rulli} C.C. Rulli and M.S. Sarandy, Phys. Rev. A {\bf 84} (2011) 042109.

\bibitem{Ma} Z-H Ma, Z-H Chen and F.F. Fanchini, New J. Phys. {\bf 15} (2013) 043023.

\bibitem{DaoudPLA} M. Daoud and R. Ahl Laamara, Phys. Lett. A {\bf 376} (2012) 2361.

\bibitem{DaoudIJQI} M. Daoud and R. Ahl Laamara, International Journal of Quantum Information {\bf 10} (2012) 1250060.


\subsection{Kitagawa-Ueda state}

We discuss an ensemble of $n$ spin-$\frac 12$ with exchange
symmetry whose dynamics is governed by the nonlinear Hamiltonian given by
$$H=\chi J_x^2 = \frac \chi 4 \sum_{i,j=1}^{n} \sigma_{i1}\otimes \sigma_{j1}$$
where $\chi$ is the coupling constant. This Hamiltonian involves all
pairwise interactions.  We assume that  the system is initially
 prepared in the product state $|n, 0\rangle=|111,...,1\rangle $. Considering its dynamic evolution,
 the evolved state at time $t$ writes
\begin{equation}
|\Psi (t)\rangle = e^{-i\chi tJ_x^2}|n, 0\rangle.
\end{equation}
Using the results obtained in \cite{Kitagawa} the following
expectation values are obtained ($\theta =2\chi t$)

\begin{eqnarray}
\langle J_x\rangle  &=&\langle J_y\rangle =0,  \label{eq:aaaa} \nonumber \\
\langle J_z\rangle  &=&-\frac n2\cos ^{n-1}\left( \frac \theta 2\right)  \nonumber  \\
\langle J_x^2\rangle  &=&n/4 \nonumber  \\
\langle J_y^2\rangle  &=&\frac 18\left( n^2+n-n(n-1)\cos
^{n-2}\theta \right)
\nonumber \\
\langle J_z^2\rangle  &=&\frac 18\left( n^2+n+n(n-1)\cos
^{n-2}\theta \right)
\nonumber \\
\langle [J_{+},J_z]_{+}\rangle  &=&0 \nonumber \\
\langle [J_x,J_y]_{+}\rangle  &=&\frac 12n(n-1)\cos
^{n-2}\left(\frac \theta 2\right)\sin \left(\frac \theta 2\right).
\label{eq:bbbb}
\end{eqnarray}




The expectation values of local spin operators are given by



$$  \langle \sigma_{i3} \otimes \sigma_{j0}\rangle = -  \cos ^{n-1}\left( \frac \theta 2\right)  , \qquad \langle \sigma_{i3} \otimes \sigma_{j3}\rangle = \frac 12 \bigg(1
+\cos ^{n-2}\theta  \bigg)$$
$$  \langle \sigma_{i+} \otimes \sigma_{j-}\rangle =  \frac 18 \bigg(1
-\cos ^{n-2}\theta  \bigg), \qquad \langle \sigma_{i-} \otimes
\sigma_{j-}\rangle =  \frac 18 \bigg(\cos ^{n-2}\theta -1
\bigg)-\frac i2 \cos ^{n-2}\left( \frac \theta 2\right) \sin \left(
\frac \theta 2\right)$$



The elements of the two-qubit reduced density matrix write


\begin{eqnarray}
\rho_{11}  &=& \frac 18 \bigg( 3 + \cos ^{n-2}\theta - 4 \cos ^{n-1}\left( \frac \theta 2\right) \bigg)  \nonumber \\
\rho_{44}   &=& \frac 18 \bigg( 3 + \cos ^{n-2}\theta + 4 \cos ^{n-1}\left( \frac \theta 2\right) \bigg)   \nonumber  \\
\rho_{41}   &=& \frac 18 \bigg(\cos ^{n-2}\theta -1 \bigg)+\frac i2 \cos ^{n-2}\left( \frac \theta 2\right) \sin \left( \frac \theta 2\right)\nonumber  \\
\rho_{22} &=& \rho_{23} = \rho_{32}= \rho_{33} = \frac 18 \bigg(1
-\cos ^{n-2}\theta  \bigg)
\end{eqnarray}



The equations (\ref{omega11}), (\ref{omega22}) and (\ref{Omega33})
lead to

\begin{equation}
\omega_{11} = \sqrt{2\rho_{22}} ~\frac{ \rho_{11} + \rho_{44}+ 2
\vert \rho_{14}\vert + 2 \sqrt{\rho_{11}\rho_{44}- \vert
\rho_{14}\vert^2}}{\sqrt{\rho_{11} + \rho_{44} + 2
\sqrt{\rho_{11}\rho_{44}- \vert \rho_{14}\vert^2}}}
\end{equation}
\begin{equation}
\omega_{22} = \sqrt{2\rho_{22}} ~\frac{ \rho_{11} + \rho_{44}- 2
\vert \rho_{14}\vert + 2 \sqrt{\rho_{11}\rho_{44}- \vert
\rho_{14}\vert^2}}{\sqrt{\rho_{11} + \rho_{44} + 2
\sqrt{\rho_{11}\rho_{44}- \vert \rho_{14}\vert^2}}}
\end{equation}
\begin{equation}
\omega_{33} = \rho_{11} + \rho_{44} - 4 \frac{\vert
\rho_{14}\vert^2}{\rho_{11} + \rho_{44} + 2
\sqrt{\rho_{11}\rho_{44}- \vert \rho_{14}\vert^2}}
\end{equation}


\bibitem{Ollivier2001} H. Ollivier and W. H. Zurek,
prl \textbf{88}, 017901 (2001).

\bibitem{Henderson2001} L. Henderson and V. Vedral,
J. Phys. A \textbf{34}, 6899 (2001).

\bibitem{Datta2008}
A. Datta, A. Shaji, and C. M. Caves, prl \textbf{100}, 050502
(2008).

\bibitem{Lanyon2008}
B. P. Lanyon, M. Barbieri, M. P. Almeida, and  A. G. White, prl
\textbf{101}, 200501 (2008).

\bibitem{Luo2008}
S. Luo, pra \textbf{77}, 042303 (2008).

\bibitem{Dillenschneider2008}
R. Dillenschneider, prb \textbf{78}, 224413 (2008).

\bibitem{Sarandy2009}
 M. S. Sarandy,
pra \textbf{80}, 022108 (2009).

\bibitem{Ali2010}
M. Ali, A. R. P. Rau, and G. Alber, pra, \textbf{81}, 042105 (2010).

\bibitem{Adesso2010}
G. Adesso and A. Datta, prl, \textbf{105}, 030501 (2010).

\bibitem{Giorda2010}
P. Giorda and M. G. A. Paris, prl, \textbf{105}, 020503 (2010).

\bibitem{YXChen2010}
Y.-X Chen and S.-W Li, pra \textbf{81}, 032120 (2010).

\bibitem{Mazzola2010}
L. Mazzola, J. Piilo, S. Maniscalco, arXiv:1001.5441v2.

\bibitem{Maziero2010}
J. Maziero, T. Werlang , F. F. Fanchini, L. C. C\'{e}leri, and R. M.
Serra, pra \textbf{81}, 022116 (2010).

\bibitem{Maziero2009}
J. Maziero, L. C. C\'{e}leri, R. M. Serra, and V. Vedral, pra
\textbf{80}, 044102 (2009).

\bibitem{Werlang2009}
 T. Werlang, S. Souza, F. F. Fanchini, and C. J. Villas Boas,
pra \textbf{80}, 024103 (2009).

\bibitem{BWang2010}
B. Wang, Z.-Y Xu, Z.-Q. Chen, and M. Feng, pra \textbf{81}, 014101
(2010).

\bibitem{Fanchini2009}
F. F. Fanchini, T. Werlang, C. A. Brasil, L. G. E. Arruda, and A. O.
Caldeira, pra \textbf{81}, 052107 (2010).

\bibitem{JSXu2010}
J.-S. Xu, X.-Y. Xu, C.-F. Li, C.-J. Zhang, X.-B. Zou, and G.-C Guo,
Nat. Commun. \textbf{1}, 7 (2010).

\bibitem{Dakic2010}
B. Daki\'{c}, V. Vedral, and \v{C}. Brukner,
  arXiv:1004.0190v1.

\bibitem{Modi2010} K. Modi, T. Paterek, W. Son, V. Vedral, and M.
  Williamson, prl \textbf{104}, 080501 (2010).

\bibitem{YXChen2010_2} Y.-X. Chen and Z. Yin, arXiv:1002.0176v1
  (2010).

\bibitem{Ferraro2009} A. Ferraro, L. Aolita, D. Cavalcanti,
  F. M. Cucchietti, and A. Acin, arXiv:0908.3157v3.

\bibitem{Werlang2010} T. Werlang and G. Rigolin, pra \textbf{81},
  044101 (2010).

\bibitem{Maziero2010_2} J. Maziero, L. C. C\'{e}leri, and R. M. Serra,
  arXiv:1004.2082v1.

\bibitem{Bylicka2010} B. Bylicka and D. Chru\'{s}ci\'{n}ski,
  arXiv:1004.0434v1.

\bibitem{Datta2010} A. Datta, eprint arXiv:1003.5256v1.

\bibitem{Datta2009}
A. Datta and S. Gharibian, pra \textbf{79}, 042325 (2009).

\bibitem{Vedral1997} V. Vedral, M. Plenio, M. Rippin, and P. Knight,
  prl \textbf{78}, 2275 (1997).

\bibitem{TCWei2003} T.-C. Wei and P. M. Goldbart, pra \textbf{68},
  042307 (2003).

\bibitem{Yu2004} T. Yu and J. H. Eberly, prl \textbf{93}, 140404
  (2004).

\bibitem{Eberly2007} J. H. Eberly and T. Yu, Science \textbf{316}, 555
  (2007).

\bibitem{Yu2009} T. Yu and J. H. Eberly, Science \textbf{323}, 598
  (2009).


\bibitem{Yu2007} T. Yu and J.H. Eberly, J. Mod. Opt. \textbf{54}, 2289
  (2007).

\bibitem{Zhou2010A} D. Zhou and R. Joynt, arXiv:1006.5474.

\bibitem{Zhou2010B} D. Zhou, G.-W. Chern, J. Fei, and R. Joynt,
  arXiv:1007.1749.

\bibitem{Koashi2004} M. Koashi and A. Winter, Phys. Rev. A
  \textbf{69}, 022309 (2004).

\bibitem{Cen2010} L.-X. Cen, X.-Q. Li, J.S. Shao, and Y.J. Yan,
  arXiv:1006.4727.

\bibitem{Groisman2005} B. Groisman, S. Popescu, and A. Winter, pra
  \textbf{72}, 032317 (2005).

\bibitem{Schlienz1995} J. Schlienz and G. Mahler, pra \textbf{52},
  4396 (1995).

\bibitem{Wang2010} X. Wang, A. Miranowicz, Y.-X Liu, C. P. Sun, and
  F. Nori, pra \textbf{81}, 022106 (2010).

\bibitem{NielsenBook} M. A. Nielsen, I. L. Chuang, \emph{Quantum
    Computation and Quantum Information,} (Cambridge University Press,
  Cambridge, UK, 2000).

\bibitem{PreskillLect} J. Preskill, \emph{Lecture Notes for Physics
    219: Quantum Information and Computation} (Caltech, Pasadena, CA,
  1999), Chap. 3,


\bibitem{Horodecki1996} R. Horodecki and M. Horodecki, pra
  \textbf{54}, 1838 (1996).

\bibitem{Hill1997} S. Hill and W. K. Wootters, prl \textbf{78}, 5022
  (1997).

\bibitem{Wootters1998} W. K. Wootters, prl \textbf{80}, 2245 (1998).


\bibitem{ref1} See, for instance, M. Nielsen and I. Chuang, {\it Quantum Computation and Quantum Information} (Cambridge University Press, New York, 2000).

\bibitem{ref2} H. Ollivier and W. H. Zurek, Phys. Rev. Lett. {\bf 88}, 017001 (2002); L. Henderson and V. Vedral, {\it ibid} {\bf 90}, 050401 (2003); J. Maziero, L. C. Celeri, R. M. Serra, and V. Vedral, arXiv: 0905.3396.

\bibitem{ref3} T. Yu and J. H. Eberly, Quantum Inform. and Comput. {\bf 7}, 459 (2007); Phys. Rev. Lett. {\bf 93}, 140404 (2004); {\it ibid} {\bf 97}, 140403 (2006).

\bibitem{ref4} See, for instance, S. Bose, I. Fuentes-Guridi, P. L. Knight, and V. Vedral, Phys. Rev. Lett. {\bf 87}, 050401 (2001); G. L, Kamta and A. F. Starace, {\it ibid} {\bf 88}, 107901 (2002); J. S. Pratt, {\it ibid} {\bf 93}, 237205 (2004); S. J. Gu, G. S. Tian, and H. Q. Lin, Phys. Rev. A {\bf 71}, 052322 (2005); J. Wang, H. Batelaan, J. Podany, and A. F. Starace, J. Phys. B {\bf 39}, 4343 (2006).

\bibitem{ref5} R. Dillenschneider, Phys. Rev. B {\bf 78}, 22413 (2008); S. Luo, Phys. Rev. A {\bf 77}, 042303 (2008); M. S. Sarandy, arXiv: 0905.1347; T. Werlang, S. Souza, F. F. Fanchini, and C. J. Villas Boas, arXiv: 0905.3376.



\bibitem{ref7} R. F. Werner, Phys. Rev. A {\bf 40}, 4277 (1989).

\bibitem{ref8} A. R. P. Rau, Phys. Rev. A {\bf 61}, 032301 (2000).

\bibitem{ref9} A. R. P. Rau, G. Selvaraj, and D. Uskov, Phys. Rev. A {\bf 71}, 062316 (2005).

\bibitem{ref10} D. Uskov and A. R. P. Rau, Phys. Rev. A {\bf 74}, 030304 (R) (2005); {\it ibid} {\bf 78}, 022331 (2008); Sai Vinjanampathy and A. R. P. Rau, arXiv: 0906.1259.

\bibitem{ref11} A. R. P. Rau, Phys. Rev. A {\bf 79}, 042323 (2009).

\bibitem{ref12} F. J. M. van de Ven and C. W. Hilbers, J. Magn. Reson. {\bf 54}, 512 (1983).

\bibitem{ref13} J. Zhang, J. Vala, S. Sastry, and K. B. Whaley, Phys. Rev. A {\bf 67}, 042313 (2003).

\bibitem{ref14} T. Beth, D. Jungnickel, and H. Lenz, {\it Design Theory}, Vols. 1 and 2, Encyclopaedia of Mathematics, Vol. 69 (Bibl. Inst., Z\"{u}rich, 1985; Cambridge University Press, 1993).

\bibitem{ref15} See, for instance, G. M. Dixon, {\it Division Algebras: Octonions, Quaternions, Complex Numbers and the Algebraic Design of Physics}, Vol. 290 of Mathematics and its Applications (Kluwer, Dordrecht, 1994); L. E. Dickson, Ann. Of Math. {\bf 20}, 155 (1919); H. S. M. Coxeter, Duke Math. J. {\bf  13}, 561 (1946); J. C. Baez, Bull. Am. Math. Soc. {\bf 39}, 145 (2002) and www.math.ucr.edu/home/baez/octonions/

\bibitem{ref16} S. Hill and W. K. Wootters, Phys. Rev. Lett. {\bf 78}, 5022 (1997); W. K. Wootters, {\it ibid} {\bf 80}, 2245 (1998).



\bibitem{sachdev} S. Sachdev, \emph{Quantum phase transitions} (Cambridge University Press, Cambridge, 2011).

\bibitem{fisher2} M. P. A. Fisher, P. B. Weichman, G. Grinstein and D. S. Fisher, Phys. Rev. B, \textbf{40}, 546 (1989).

\bibitem{liu2} Y. Liu and A. M. Goldman, Physica D \textbf{83}, 163 (1995).

\bibitem{mathur} N. D. Mathur, F. M. Grosche, S. R. Julian, I. R. Walker, D. M. Freye, R. K. W. Haselwimmer and G. G. Lonzarich, Nature \textbf{394}, 39 (1998).

\bibitem{Lohneysen07} H. V. L\"{o}hneysen, A. Rosh, M. Vojta and W\"{o}lfle, Rev. Mod. Phys. \textbf{79}, 1015 (2007).

\bibitem{sachdev2} S. Sachdev, Phys. World \textbf{10}, 25 (1994).

\bibitem{osterloh} A. Osterloh, L. Amico, G. Falci and R. Fazio, Nature \textbf{416}, 608 (2002).

\bibitem{osborne} T. J. Osborne and M. A. Nielsen, Phys. Rev. A \textbf{66}, 032110 (2002).

\bibitem{amicoRev} L. Amico, R. Fazio, A. Osterloh and V. Vedral, Rev. Mod. Phys. \textbf{80}, 517 (2008).

\bibitem{sarandy1} M. S. Sarandy, Phys. Rev. A \textbf{80}, 022108 (2009).

\bibitem{maziero} J. Maziero, H. C. Guzman, L. C. C\'{e}leri, M. S. Sarandy and R. M. Serra, Phys. Rev. A \textbf{82}, 012106 (2010).

\bibitem{maziero1} J. Maziero, L. C. C\'{e}leri, R. M. Serra and M. Sarandy, Phys. Lett. A \textbf{376}, 1540 (2012).

\bibitem{sarandy3} M. S. Sarandy, T. R. de Oliveira and L. Amico, Int. J. Mod. Phys. B \textbf{27}, (2013).

\bibitem{werlang} T. Werlang, C. Trippe, G. A. P. Ribeiro and G. Rigolin, Phys. Rev. Lett. \textbf{105}, 095702 (2010).

\bibitem{werlang1} T. Werlang, G. A. P. Ribeiro and G. Rigolin, Phys. Rev. A \textbf{83}, 062334 (2011).

\bibitem{sarandy2} C. C. Rulli and M. S. Sarandy, Phys. Rev. A \textbf{84}, 042109 (2011).

\bibitem{justino} L. Justino and T. R. de Oliveira, Phys. Rev. A \textbf{85}, 052128 (2012).

\bibitem{cheng} W. W. Cheng, C. J. Shan, Y. B. Sheng, L. Y. Gong, S. M. Zhao and B. Y. Zheng, Physica E \textbf{44}, 1320 (2012).

\bibitem{shan} C.-J. Shan, W.-W. Cheng, J.-B. Liu, Y.-S. Cheng and T.-K. Liu, Sci. Rep. \text{4}, 4473 (2014).

\bibitem{ollivier} H. Ollivier and W. H. Zurek, Phys. Rev. Lett. \textbf{88}, 017901 (2001).

\bibitem{vedral} L. Henderson and V. Vedral, J. Phys. A: Math. Gen. \textbf{34}, 6899 (2001).

 \bibitem{mile} M. Gu, H. M. Chrzanowski, S. M. Assad, T. Symul, K. Modi, T. C. Ralph, V. Vedral and P. K. Lam, Nature Phys. \textbf{8}, 671 (2012)

\bibitem{philip} B. Daki\'{c}, Y. O. Lipp, X. Ma, M. Ringbauer, S. Kropastschek, S. Barz, T. Paterek, V. Vedral, A. Zeilinger, \u{C}. Brukner and P. Walther, Nature Physics \textbf{8}, 666 (2012).

\bibitem{datta08} A. Datta, A. Shaji and C. M. Caves, Phys. Rev. Lett. \textbf{100}, 050502 (2008).

\bibitem{lanyon08} B. P. Lanyon, M. Barbieri, M. P. Almeida and A. G. White, Phys. Rev. Lett. \textbf{101}, 200501 (2008).

\bibitem{modi} K. Modi, A. Brodutch, H. Cable, T. Paterek and V. Vedral, Rev. Mod. Phys. \textbf{84}, 1655 (2012).



\bibitem{suzuki} M. Suzuki, Prog. Theor. Phys. \textbf{46}, 1337 (1971).

\bibitem{titvinidze} I. Titvinidze and G. I. Japaridze, Eur. Phys. J. B \textbf{32}, 383 (2003).

\bibitem{li} Y-C. Li and H-Q. Lin, Phys. Rev. A \textbf{83}, 052323 (2011).

\bibitem{anderson} P. W. Anderson, Mater. Res. Bull. \textbf{8}, 153 (1973).

\bibitem{balents} L. Balents, Nature \textbf{464}, 199 (2010).

\bibitem{fanchini} G. Karpat, B. \c{C}akmak and F. F. Fanchini, Phys. Rev. B \textbf{90}, 104431 (2014).

\bibitem{Zvyagin} A. A. Zvyagin, Phys. Rev. B \textbf{80}, 014414 (2009).

\bibitem{sudden1} J. Maziero, L. C. C\'{e}leri, R. M. Serra and V. Vedral, Phys. Rev. A \textbf{80}, 044102 (2009).

\bibitem{sudden2} R. Auccaise, L. C. C\'{e}leri, D. O. Soares-Pinto, E. R. de Azevedo, J. Maziero, A. M. Souza, T. J. Bonagamba, R. S. Sarthour, I. S. Oliveira and R. M. Serra, Phys. Rev. Lett. \textbf{107}, 140403 (2011).

\bibitem{giovannetti} V. Giovannetti, S. Lloyd and L. Maccone, Nature Photonics \textbf{5}, 222 (2011).

\bibitem{marzolino} U. Marzolino and T. Prosen, Phys. Rev. A \textbf{90}, 062130 (2014).

\bibitem{lucas} K. Micadei, R. M. Serra and L. C. C\'{e}leri, Phys. Rev. E \textbf{88}, 062123 (2013).

\bibitem{lutz} S. Deffner and E. Lutz, Phys. Rev. Lett. \textbf{107}, 140404 (2011).

\bibitem{john} E. Mascarenhas, H. Braganca, R. Dorner, M. Franca Santos, V. Vedral, K. Modi and J. Goold, Phys. Rev. E \textbf{89}, 062103 (2014).


